Bank of England

The gravity of syndication ties in international equity underwriting

Staff Working Paper No. 1,021

April 2023

Luke Heath Milsom, Vladimír Pažitka, Isabelle Roland and Dariusz Wójcik

Staff Working Papers describe research in progress by the author(s) and are published to elicit comments and to further debate. Any views expressed are solely those of the author(s) and so cannot be taken to represent those of the Bank of England or to state Bank of England policy. This paper should therefore not be reported as representing the views of the Bank of England or members of the Monetary Policy Committee, Financial Policy Committee or Prudential Regulation Committee.



Bank of England

Staff Working Paper No. 1,021

The gravity of syndication ties in international equity underwriting

Luke Heath Milsom,⁽¹⁾ Vladimír Pažitka,⁽²⁾ Isabelle Roland⁽³⁾ and Dariusz Wójcik⁽⁴⁾

Abstract

We examine how cross-border syndication ties reduce information frictions and positively impact exports of equity underwriting services. Using a panel data set from 2000–15, we develop a measure of information flows based on 'core syndication ties' where the lead underwriter is in either the importing or exporting country. We find that new core syndication ties have a significant and positive effect on exports. This finding is supported by evidence from 'peripheral syndication ties', which are associated with smaller information flows, and an instrumental variable approach which focuses on plausibly exogenous supply-side shocks. Furthermore, the effect of new core syndication ties is stronger when information frictions between trading partners are more severe and for more information-sensitive transactions like IPOs.

Key words: Gravity, international trade, international finance, equity securities underwriting, cross-border bank linkages, financial geography.

JEL classification: F14, F23, F36, F65, G15, G24.

- (2) University of Leeds, Leeds University Business School. Email: v.pazitka@leeds.ac.uk
- (3) Bank of England. Email: isabelle.roland@bankofengland.co.uk
- (4) University of Oxford, School of Geography and the Environment. Email: dariusz.wojcik@spc.ox.ac.uk

The views expressed in this paper are those of the authors, and not necessarily those of the Bank of England or its committees. We thank João Santos Silva and Silvana Tenreyro for helpful advice with the use of PPML, and Yoto Yotov for comments and suggestions. We also thank two anonymous referees for invaluable comments and suggestions. The project has received funding from the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation program (grant agreement number 681337). The article reflects only the authors' views and the ERC is not responsible for any use that may be made of the information it contains. The funding source had no involvement in study design, in the collection, analysis and interpretation of data, in the writing of the report, and in the decision to submit the article for publication.

⁽¹⁾ University of Oxford, Department of Economics. Email: luke.milsom@economics.ox.ac.uk

The Bank's working paper series can be found at www.bankofengland.co.uk/working-paper/staff-working-papers

Bank of England, Threadneedle Street, London, EC2R 8AH Email: enquiries@bankofengland.co.uk

©2023 Bank of England ISSN 1749-9135 (on-line)

1 Introduction

International bank linkages are a key feature of global financial market integration. In equity underwriting services, cross-border underwriting syndicates are a driving force for integration. Syndicates are groups of underwriters organized for the purpose of bringing a single issue of equity securities to the market and dissolved thereafter (Pichler and Wilhelm, 2001). Typically, a lead underwriter¹ manages the relationship with the issuer and is primarily responsible for due diligence. The other syndicate members sell securities to investors within their networks and may contribute equity analyst coverage of the issuer (Ljungqvist et al., 2009). Underwriters involved in a syndicate collectively have a higher potential for information collection and dissemination than any individual syndicate member. Ties formed through international underwriting syndicates therefore facilitate information flows and mitigate information frictions between issuers and investors located in different countries. We argue that, through this channel, syndication ties promote trade in equity underwriting services, and provide supporting evidence using theory-consistent gravity equations.

We build a measure of newly formed "core syndication ties" that proxies for information flows in either direction between the trading partners. To do so, we exploit the structure of underwriting syndicates. The lead underwriter is primarily responsible for conducting due diligence on the issuer and uses other syndicate members to produce and disseminate information about the issuer and investor demand for the offering. Therefore, information flows between the trading partners are strongest when the underwriter in the importing (issuing) or exporting country is the lead underwriter. We denote syndication ties where the underwriter in either the importing or exporting country is the lead underwriter as "core syndication ties". While information is still expected to flow between the trading partners when the underwriters in both countries are non-lead underwriters, the information content of new syndication ties should be lower. We denote syndication ties where the underwriters in both countries are non-lead underwriters as "peripheral syndication ties". Separating newly formed syndication ties into two categories based on the strength of the information flows they generate enables us to provide evidence for the information channel. For each category of syndication ties, the stock of ties in year t is measured using the count of unique bank dyads that are connected through co-membership of underwriting syndicates for each pair of countries up to year t. The question we ask is whether newly formed ties increase the flow of underwriting services between trading partners. Therefore, our explanatory variables measure *changes* in syndication ties over time. More specifically, they measure the number of newly connected bank dyads for each pair of countries in each year.

Our data set addresses the scarcity of disaggregated data on trade in financial services. We construct a country-pair panel data set using transaction-level data on fees earned by banks on equity underwriting services. Specifically, we compile a comprehensive data set of trade flows by aggregating transaction-level data on revenue flows associated with the underwriting of new (primary and secondary) issues of equity securities from the Dealogic Equity Capital Market (ECM) database. Our

¹The lead underwriter is often referred to as bookrunner.

data set covers 91,511 transactions from 7,326 underwriters serving 46,408 equity issuers across 122 countries of origin and 145 countries of destination for the period 2000-2015. In line with best practice in the presence of many zero trade flows, the gravity equations are estimated with Poisson pseudo maximum likelihood (PPML). The causal interpretation of our baseline results is potentially threatened by unobserved common factors that drive both syndication ties and exports of equity underwriting services. To address the presence of common factors, we saturate our regression models with fixed effects, namely importer-time, exporter-time, and country-pair fixed effects.

We begin our empirical analysis by testing whether new core syndication ties formed between exporting country j and importing (issuing) country k between times t - 1 and t affect exports from j to k at time t. We find that the effect is statistically and economically significant. Specifically, a 1% increase in new core syndication ties increases exports by 0.243%. In addition, we show that the effect displays some persistence. Core syndication ties formed between t - 2 and t - 1 also have a significant effect on exports of equity underwriting services. The fact that lagged changes in syndication ties are positively correlated with flows of equity underwriting services is suggestive of causality. The causal interpretation of our baseline result, however, is potentially threatened by reverse causality. Identification is endangered by the fact that underwriters from export markets might establish new linkages in import markets in anticipation of an increase in exports due to demand shocks in the importing country. We address the potential for reverse causality with a "placebo test" and an instrumental variable (IV) approach.

Our first test is akin to a placebo test. We test whether new peripheral syndication ties (formed by non-lead underwriters in countries k and j) affect exports from j to k less strongly than new core syndication ties. As explained earlier, we expect new syndication ties between non-lead underwriters to be associated with smaller information flows, so we should find a smaller effect on exports. Indeed, information acquisition is mainly conducted by lead underwriters. This analysis serves as a placebo test: if our benchmark result was purely driven by spurious correlation between new syndication ties and exports of equity underwriting services, it would be as likely to show up in the coefficient on the most informative (core) and the least informative (peripheral) syndication ties. In addition, the test provides the first piece of evidence in support of our hypothesis that underwriting syndicates promote exports through information production. We find that the effect of a 1% increase in new core syndication ties on exports is roughly seven times larger than the effect of newly formed peripheral ties.

The result of this test places limits on the extent to which reverse causality might be driving our findings. To further rule out endogeneity concerns, we develop an instrumental variable (IV) approach. We identify supply-side shocks to new core syndication ties between exporting country j and importing country k by measuring shifts in country j's underwriters' interest in country k as an export destination unrelated to changes in demand-side prospects in country k. In particular, we instrument newly formed core ties between j and k with newly formed core ties between j and i. The subscript i denotes a country that is comparable to country k as an export market for country j's underwriters.

The idea is that newly formed core ties between j and i measure country i's attractiveness to country j's underwriters, which is plausibly exogenous to demand-side shocks to exports from j to k. Our relevance assumption is that positive shocks to the attractiveness of country i (newly formed core ties between j and i) constitute negative shocks to the attractiveness of country k. Our exogeneity assumption is that newly formed core syndication ties between j and i only affect exports from j to k through their impact on newly formed core ties between j and k. Our baseline results are validated. We find that a 1% increase in new core syndication ties increases exports by 0.221% (compared to 0.243% when new core ties are not instrumented).

Having established solid evidence for our baseline result, we then proceed to study the mechanism in more detail. The literature on gravity in international finance provides evidence on the role of information frictions in cross-border transactions in financial assets (e.g. Portes et al., 2001; Portes and Rey, 2005). The main information frictions affecting trade in equity underwriting services relate to the barriers that investors face in accessing and interpreting price relevant information, particularly soft information, about foreign issuers. Higher information barriers between the issuer's country (importer) and the underwriter's country (exporter) lead to a higher cost of capital because investors demand a higher risk premium as compensation for information asymmetry. Given the negative relationship between bilateral flows in equity underwriting services and the cost of capital faced by individual issuers, information frictions are expected to reduce trade. Information frictions can, however, be mitigated by financial and reputational intermediaries, including securities underwriters' networks. We review the relevant literature in Sections 2.1 and 2.4.

We provide evidence that underwriting syndicates promote exports through information production. The first piece of evidence is the placebo test described above. In addition to alleviating concerns about spurious correlation between exports of equity underwriting services and new syndication ties, our placebo test shows that changes in syndication ties that generate large information flows between the trading partners (core ties) matter more for exports than ties that generate smaller information flows (peripheral ties). We further establish the plausibility of the information mechanism using two approaches that show that newly formed core syndication ties have a stronger effect on exports of equity underwriting services when information frictions between the trading partners are more severe. In our first approach, we first use OECD membership as a crude proxy for contract enforcement. We find that the effect of new core syndication linkages on exports is larger when the exporter is an OECD member but the importer is not. Second, we find that the effect of new core syndication ties on exports is larger when the destination country is riskier. The risk proxies include the importing country's International Country Risk Guide index, sovereign credit rating (as measured by Moody's), and Resolving Insolvency score from the World Bank Doing Business survey (overall score and recovery rate score). For the second approach, we demonstrate that the effect of new core syndication ties on exports is larger for more information-sensitive transactions. Specifically, we show that the effect is larger for initial public offerings (IPOs) than for other transactions (follow-on offerings and convertible debt) - the former being more information sensitive.

The remainder of the paper proceeds as follows. Section 2 briefly describes the related literature and highlights our novel contributions. Section 3 describes the empirical gravity model, its application to trade in equity underwriting services, and its empirical implementation. Section 4 describes our data. In Section 5 we present and discuss our results. Section 6 concludes.

2 Related literature and novel contributions

2.1 Gravity in international finance

First, our paper is related to the literature on gravity in international finance. The interest in gravity was sparked by the finding that gravity equations explain international transactions in financial assets at least as well as trade transactions for goods (Portes et al., 2001; Portes and Rey, 2005). Since then a large number of papers have estimated gravity equations for cross-border equity, bond and bank holdings.² In contrast with this literature, we focus on financial services provided by banks in the context of the issuance of equity securities rather than on the security holdings themselves. Focusing on services rather than asset holdings offers the advantage that we escape the criticism of Okawa and van Wincoop (2012) that applies to most of the literature on gravity in international finance. Most contributions are in violation of the theory of gravity because they include explanatory variables without theoretical foundation. According to the theory of optimal portfolio choice, the demand for an asset depends not only on its expected return, but also on the latter's covariance with other assets. As a consequence, researchers often include correlations (mainly across asset returns) as explanatory variables in their gravity equations.³ However, this is not consistent with gravity theory. Indeed, one of the key assumptions that must be satisfied to generate a gravity specification for asset holdings is that asset demand depends on a relative price. This assumption is problematic because it is not in line with the theory of optimal portfolio choice. Our paper is not subject to this caveat.

The literature on gravity in international finance provides evidence on the role of information frictions in cross-border transactions in financial assets (e.g. Portes et al., 2001; Portes and Rey, 2005). According to Portes and Rey (2005), informational frictions are the main factor shaping the geographical distribution of international equity transactions. The role of information barriers in cross-border asset holdings and transactions is reflected in a similar finding in the literature on overseas listings. In-

²Papers that examine cross-border bank holdings using the BIS *Locational Banking Statistics* include among others Papaioannou (2009), Brei and von Peter (2018), Daude and Fratzscher (2008), and Houston, Lin and Ma (2012). Papers that examine equity holdings using the IMF *Coordinated Portfolio Investment Survey* (CPIS) include among others Lane and Milesi-Ferretti (2008), Daude and Fratzscher (2008), Lane and Milesi-Ferretti (2005), Veronique and Benassy-Quere (2006), Faruqee et al. (2004), Coeurdacier and Martin (2009), Coeurdacier and Guibaud (2011), and Berkel (2007). The literature that relies on other data sets is dominated by papers focusing primarily on a single country, with most attention paid to the United States; e.g. Portes et al. (2001), Ahearne et al. (2004), Forbes (2010), and Chitu et al. (2014).

³See among others Lane and Milesi-Ferretti (2008), Forbes (2010), Faruqee et al. (2004), Coeurdacier and Guibaud (2011), Berkel (2007), Portes and Rey (2005), and Chan et al. (2005).

deed, the results of Sarkissian and Schill (2004) on proximity and familiarity preference in the choice of listing venues indicate that asymmetric information plays an important role in issuers' choice of overseas listing venues. The main information frictions affecting trade in equity underwriting services relate to the barriers that investors face in accessing and interpreting price relevant information, particularly soft information, about foreign issuers. Higher information barriers between the issuer's country (importer) and the underwriter's country (exporter) lead to a higher cost of capital because investors demand a higher risk premium as compensation for information asymmetry. Given the negative relationship between bilateral flows in equity underwriting services and the cost of capital faced by individual issuers, information frictions are expected to reduce trade. Information frictions can, however, be mitigated by financial and reputational intermediaries, including venture capitalists, equity research analysts, accountants, auditors, law firms and securities underwriters among others (see e.g. Dunbar, 2000; Pollock et al., 2004; Ljungqvist et al., 2009; Jeon and Ligon, 2011). We focus on the role of securities underwriters' networks.

2.2 Gravity in services trade

Second, our paper is related to the literature on gravity in services (see e.g. Kimura and Lee, 2006; Anderson et al., 2018). The paucity of comparable cross-country, time series data on services means that the empirical literature is still sparse. Data sets on trade in services are compiled on the basis of balance of payments data provided by national statistics offices. There are many challenges to overcome in order to develop a global database of coherent bilateral trade in services statistics. These challenges include missing data, trade asymmetries (where the reported exports of country j to country k do not align with the reported imports of country k from country j), and differences in reporting standards and statistical methodologies. These challenges are particularly acute when trying to compile a database at a disaggregated level to look at certain categories of services in isolation. Due to severe data limitations, gravity estimates for services are still often based on total trade with the rest of the world or are limited to a small number of (mostly high-income) countries⁴. Contributions that focus specifically on trade in financial services are even scarcer (see e.g. Nordås and Rouzet, 2017; Brei and von Peter, 2018). Brei and von Peter (2018) use the BIS Locational Banking Statistics to estimate theory-consistent gravity equations for international banking. The authors argue that if information frictions matter, lending should be greater when banks have better access to information on foreign markets. They condition the amount of credit extended from a country to another on whether banks from the country of origin have branches or subsidiaries operating in the country of destination. The presence of affiliates abroad enters with a large positive coefficient.

Our paper contributes to this literature in several ways. We construct an entirely novel panel data set which presents numerous advantages over data sets that are based on balance of payments statistics. First, our data set has a comprehensive global coverage (zero observations are true zeros with a very

⁴Contributions include Kimura and Lee (2006), Francois and Wignaraja (2008), Ceglowski (2006), Head et al. (2009), Aichele et al. (2014), and Anderson et al. (2018).

high probability) and does not require any manipulations designed to harmonize the data across countries, to mirror exports and imports, and to predict missing trade flows among countries. In particular, using "synthetic" observations obtained through mirroring and prediction techniques might be problematic in regressions that study the determinants of bilateral trade because there is a danger that a similar gravity model was used to fill in the missing cells in the trade matrix. In addition, the amount of modeling contained in the data will vary across country groups. The data set might be heavy on modeling for emerging markets for example, particularly when applying sectoral disaggregation.

Second, our data set enables us to zoom in to a very specific type of financial service. Francois and Hoekman (2010) highlight that it "makes little sense to speak of 'the service sector'." Indeed, services differ significantly in terms of their roles in the economy, their market structures, and the relative importance of the various channels through which firms can access foreign markets. In this paper, we heed their advice and zoom in on equity securities underwriting. In 2015, \$5.25 trillion of services were exported globally, out of which \$457.81 billion, approximately 8.7%, were exports of financial services (UNCTADSTAT, 2021). Exports of equity underwriting services were \$7.80 billion in 2015 (Dealogic ECM database), approximately 1.7% of all exports of financial services in the same year. Although trade in equity underwriting services is still a relatively small percentage of world GDP, it plays a crucial role in enabling firms to access global capital markets.

Third, our raw data is at the transaction level, which enables us to construct measures of crossborder connections specific to syndication activity in the equity underwriting industry. This allows us to contribute to the research agenda on *modes of supply* in services trade (see e.g. Francois and Hoekman, 2010). Providers of financial services in one country can compete with providers in other countries through four principal modes of supply (see e.g. Sampson and Snape, 1985; Webster and Hardwick, 2005). In the case of financial services, the most significant modes of supply are crossborder exporting of financial services and the establishment of a commercial presence in the importing country. We study another mode of supply which is specific to the underwriting of equity securities, namely co-membership of underwriting syndicates. The literature on modes of supply is sparse because of a dearth of data on sales of services by foreign affiliates of multinational enterprises (Francois and Hoekman, 2010).

2.3 Cross-border bank linkages and trade in goods

There is a growing body of literature on the relationship between finance and trade in goods.⁵ Recent papers highlight the role of international bank linkages in promoting trade in goods through various channels. Claessens and Van Horen (2021) examine the link between the presence of foreign banks in the exporting country and international exports of goods. They find that exports are larger when a foreign bank from the importing country is present in the exporting country. The channel is that the presence of foreign banks reduces financial frictions faced by exporting firms. Caballero et al. (2018)

⁵See surveys by Manova and Foley (2015) and Contessi and de Nicola (2012).

also study the link between international trade in goods and bank linkages, but measure linkages differently. Bank linkages are measured as the number of bank pairs in the exporting and importing countries that are connected through syndicated lending to each other. Their results indicate that not only do bank linkages increase exports of goods, they do so by reducing export risk.

Our paper is related to these contributions in that we study the role of international bank linkages in trade. However, our contribution is different in many ways. First, unlike Claessens and Van Horen (2021) and Caballero et al. (2018), we study trade in financial services as opposed to trade in goods. Contributions on financial services are scarce, especially at a disaggregated level such as the equity underwriting services industry. We argue that international bank linkages matter for trade in financial services, too, through channels that might be different than the channels that matter for trade in goods. Second, our measures of international bank linkages are tailored to the equity underwriting industry due to the fact that we zoom in on the latter. Specifically, we focus on newly formed syndication ties because cross-border underwriting syndicates are a specific feature (or mode of supply) in the industry we study. We exploit the hierarchical structure of equity underwriting syndicates to build a proxy for information flows between the trading partners. The measures of bank linkages in Claessens and Van Horen (2021) and Caballero et al. (2018) are entirely different. The measure of bank linkages in Claessens and Van Horen (2021) is based on the physical presence of foreign banks in the exporting country. The measure of Caballero et al. (2018) is much closer in spirit to our measure in the sense that it is a proxy for information flows that does not rely on physical presence abroad. Caballero et al. (2018) proxy information flows using data on linkages formed by bank loans extended by banks in the exporting country to banks in the importing country. Our measure of newly formed syndication ties is also a proxy for information flows but its construction is dictated by the idiosyncrasies of the syndication process (see Section 4.3). Finally, the mechanism we examine is specific to the equity underwriting industry. We propose that syndicates alleviate information frictions between underwriters in the exporting country and issuers in the importing country. We provide more information on this channel in the next section.

2.4 The role of syndicates in alleviating information frictions

An offering can be underwritten by a single underwriter or a group of underwriters, a so-called underwriting syndicate. Syndicates are groups of underwriters organized for the purpose of bringing a single issue of equity securities to the market and dissolved thereafter (Pichler and Wilhelm, 2001). There is, however, a great deal of persistence in the membership of underwriting syndicates. Therefore, recently observed syndication ties are informative about the structure of relational ties and information flows within syndication networks (Pollock et al., 2004). The purpose of underwriting syndicates goes far beyond risk sharing among underwriters. In fact, information production has been argued to be their key purpose, particularly since equity underwriters started becoming publicly listed, thereby alleviating concerns around financial capacity and the need for risk sharing for smaller privately-owned investment banks (Corwin and Schultz, 2005). Information exchange between issuers and investors is hindered by information asymmetry, which leads to investors discounting issuers' equity valuations. Reputable financial intermediaries can reduce information asymmetry by performing credible due diligence on issuers and feeding information on investor demand back to issuers (Pichler and Wilhelm, 2001). Underwriters generally face internal limitations when performing these functions, including a limited pool of investors in their respective business networks and limited internal expertise in equity valuation. Equity underwriting syndicates can alleviate information asymmetry more effectively than any individual underwriter by pooling their investor and client networks as well as their internal valuation expertise (Corwin and Schultz, 2005; Chuluun, 2015).

Typically, a lead underwriter, often referred to as bookrunner, manages the relationship with the issuer and holds primary responsibility for carrying out due diligence on the latter. Other syndicate members, including co-managers and underwriters, are primarily tasked with selling securities to investors within their networks and gathering information on investor interest in the offering (Pollock et al., 2004). However, they may also contribute to due diligence and the pricing of the offering, particularly if they conduct equity analyst research on the issuer (Ljungqvist et al., 2009). This means that underwriting syndicates are able to facilitate access to information on investor demand across a much broader group of investors than any individual underwriter (Shipilov, 2006; Jeon and Ligon, 2011; Vithanage et al., 2016) and at the same time perform superior due diligence by having collectively a higher capacity for information production (Corwin and Schultz, 2005; Ljungqvist et al., 2009; Chuluun, 2015). This consequently allows underwriting syndicates to further reduce information asymmetry between issuers and investors, which is particularly valuable in conditions of high information asymmetry such as IPOs and international offerings involving issuers and investors located in different countries. The literature therefore clearly indicates that syndication ties can proxy for information flows (Pichler and Wilhelm, 2001; Corwin and Schultz, 2005; Chuluun, 2015).

Accordingly, we use newly formed syndication ties to construct proxies for information flows across country dyads. In doing so, we take full advantage of the information embedded in the structure of underwriting syndicates. Lead underwriters act as the epicentre of every underwriting syndicate and the primary point of contact for issuers. Consequently, it is safe to assume that any key information flows between issuers and investors facilitated by an underwriting syndicate will be routed through lead underwriters (Pollock et al., 2004). For example, all the underwriters involved in a syndicate market the offering and collect information about investor demand across their respective networks of investors. This information is then passed on to the lead underwriter to inform deal pricing (Corwin and Schultz, 2005). Another specific example of information flows relates to the contribution of individual co-managers to due diligence on the issuer. Lead underwriters routinely invite other banks to join a syndicate as co-managers because the latter provide analyst coverage of the issuer and can therefore contribute their specialist expertise to due diligence and deal pricing (Ljungqvist et al., 2009). Any information contributed by individual co-managers is compiled by the lead underwriter, who holds the ultimate responsibility for due diligence, prepares the IPO prospectus summarising its findings, and communicates this to investors. Therefore, the intensity of information flows facilitated by underwriting syndicates arguably depends on whether a particular syndication tie involves a lead underwriter (core ties) or whether it connects two non-lead underwriters (peripheral ties). While core syndication ties carry prime information content, peripheral ties correspond to lesser information flows.

Our paper contributes to the body of evidence on the information acquisition role of underwriting syndicates using a novel framework in the field (gravity) by showing that various proxies for information frictions are associated with lower exports and that such frictions are reduced through syndication linkages between countries.

3 Empirical methodology

One of the most attractive features of the gravity model of trade is that it can be theoretically derived from a large class of models and be estimated separately for each industry. For example, Arkolakis et al. (2012) show that a large number of micro foundations generate isomorphic gravity equations. As described in Section 2.2, the gravity framework works very well with services data. Theoretically, Okawa and van Wincoop (2012) highlight that two key assumptions are needed to generate a gravity specification. First, total production and expenditure must be separable from the bilateral allocation of trade across countries. Second, demand must depend on a relative price, such as the price of services from a particular country relative to an overall price index. Both conditions are reasonably met for financial services trade, including trade in equity underwriting services. Trade in equity underwriting services is, however, an atypical market compared to those that are traditionally explained by gravity frameworks. Indeed, the market is relatively concentrated, implying that frameworks which emphasize perfect competition are less empirically justifiable. The market for underwriting services is however global in nature and therefore unlikely to exhibit oligopolistic tendencies⁶. Therefore, we base our application of the gravity framework on the monopolistic competition model developed in Krugman (1980), allowing for imperfect competition and potentially high entry costs as in Melitz (2003). Finally, note that the gravity framework has been used to study international banking, an industry which is also highly concentrated and global in nature (see e.g., Nordås and Rouzet, 2017; Brei and von Peter, 2018).

A traditional gravity equation that explains flows of equity underwriting services between exporting country j and importing country k at time t, X_{jkt} , is given by:

$$X_{jkt} = \exp\left\{\pi_{jt} + p_{kt} + Z_{jk}\Gamma\right\} + \varepsilon_{jkt} \tag{1}$$

 $^{^{6}\}mathrm{See}$ e.g. Papaioannou and Karagozoglu (2017) for details on market structure, concentration and competition in the underwriting industry.

where π_{jt} are exporter-time fixed effects, p_{kt} are importer-time fixed effects, and Z_{jk} is a vector of time-invariant bilateral trade costs. ε_{jkt} is an error term. The exporter-time and importer-time fixed effects account for the unobserved multilateral resistance terms in theoretical gravity equations. These multilateral resistance terms capture j's outward resistance or ease of market access, and k's inward resistance or ease of market access, respectively. Baldwin and Taglioni (2006) show the importance of sufficiently accounting for these terms in order to achieve consistent estimation of the parameters in a gravity model. The country-year fixed effects also control for any country-specific time-varying market conditions that might have an impact on issuance activity. This is important since our sample period includes data posterior to the global financial crisis. The post-crisis period saw important changes in regulation, with potential repercussions for secondary market liquidity⁷.

The traditional proxies for bilateral trade costs are the logarithm of physical distance between the trading partners $(\ln(\text{Distance}_{ik}))$, a dummy variable equal to one when the trading partners share a language (Shared language_{ik}), a dummy variable equal to one when the trading partners share a geographical border (Shared border_{ik}), and a dummy variable equal to one when the trading partners share a colonial history (Shared colonial history $_{ik}$). These variables are well suited to capture the familiarity and proximity preference in overseas listings.⁸ The literature indicates that firms that list overseas have a strong tendency to choose foreign markets that are similar to their respective domestic markets. Using a cross-section of the entire universe of overseas listings across world markets, Sarkissian and Schill (2004) find that geographic, economic, cultural, and industrial proximity play a dominant role in the choice of overseas listing venue. In other words, the choice of overseas listing venues reflects the same bias toward more proximate or familiar markets as the one found in the literature that examines investors' asset holdings.⁹ As a consequence, foreign listings are more common across markets for which diversification gains are relatively low, contrary to the idea that firms maximize international portfolio diversification gains when listing overseas.¹⁰ The preference for proximate or familiar markets is echoed by the home bias in the provision of investment banking services, although at the same time, the industry is highly internationalized and cross-border deals are very common (Wójcik, 2011). Wójcik et al. (2018) document that a fairly steady fraction of approximately 45% of global investment banking revenue was associated with cross-border deals during 2000-2015. This is comparable to the figures documented in this study. On a revenue weighted basis, 42.8% of our measured trade flows are cross-border, while the remainder are domestic.¹¹ It is noteworthy that the home bias varies significantly across countries. For example, approximately 75-80% of fees are from

⁷See Rischen and Theissen (2021) for evidence on the euro area bond market.

⁸See Karolyi (2006) for a review of the overseas listings literature.

⁹See e.g. French and Poterba (1991), Tesar and Werner (1995), Coval and Moskowitz (1999, 2001), Huberman (2001), Grinblatt and Keloharju (2001), Portes and Rey (2005), Van Nieuwerburgh and Veldkamp (2009).

¹⁰This reflects a similar finding for asset holdings: Investors neglect fundamental principles of portfolio diversification by favoring proximate assets and markets (Sarkissian and Schill, 2004). Portes and Rey (2005) provide evidence of this in a gravity framework. Once they control for informational frictions, the authors find weak support for a diversification motive in cross-border equity flows.

¹¹Domestic (cross-border) flows are defined as those where the underwriter and client (issuer) are headquartered in the same country (different countries).

cross-border flows for EMEA, 20-35% for Asia-Pacific and 20-30% for the Americas (Wójcik et al., 2018). The traditional trade costs variables included in gravity models lend themselves perfectly to capture this bias toward more proximate or familiar markets. These variables are described in Section 4.2.

Equation (1) is estimated with PPML. Trade data, especially when disaggregated by sector, suffers from the many-zeros problem. Many countries have no trade with many other countries in many time periods. Therefore, the full trade matrix is a sparse matrix containing many zero entries. This is a serious problem when using the traditional log-linearized version of the gravity equation estimated with OLS. Indeed, zeros are dropped as if they were missing values, despite the fact that they contain useful information about trade patterns. First-generation solutions to this problem, such as adding a small positive number to each zero, are arbitrary and lead to biased results (Head and Mayer, 2014). Therefore, we follow Santos Silva and Tenreyro (2006) and estimate our gravity equation using PPML¹², which allows us to retain the zeros. This method has been shown to have excellent properties, even when the proportion of zeros is very high. In addition, log-linearized models are biased (and inconsistent) in the presence of sufficient heteroskedasticity (as is the norm with trade data)¹³. Using PPML also enables us to overcome this problem as described in Santos Silva and Tenreyro (2006). For reference, we also provide results estimated with OLS.

While Equation (1) serves to illustrate that gravity is a valid framework for equity underwriting services, it is not our preferred specification. Our benchmark specification saturates the model with fixed effects by adding country-pair fixed effects and controls for changes in syndication ties. Our baseline empirical gravity equation is given by:

$$X_{jkt} = \exp\left\{\beta s_{jkt}^{\text{core}} + \pi_{jt} + p_{kt} + \rho_{jk}\right\} + \varepsilon_{jkt}$$
⁽²⁾

where π_{jt} are exporter-time fixed effects, p_{kt} are importer-time fixed effects, ρ_{jk} are country-pair fixed effects, and ε_{jkt} is an error term. The country-pair fixed effects capture all time-invariant trade costs between the trading partners and therefore absorb the traditional proxies for time-invariant bilateral trade costs. The variable of interest, s_{jkt}^{core} , measures the change in cross-border core syndication ties between the importing and exporting countries between times t - 1 and t. To obtain this measure, we first compute the cumulative sum of unique linkages formed by all syndicate relationships between countries j and k where the lead underwriter is located in either j or k (S_{jkt}^{core}). We then take the log transformation of the one-period change in the aggregate number of core ties, which is the natural

¹²PPML is a weighted non-linear least squares estimator where the weights are constructed under the assumption that $\mathbb{E}[y|x] \propto \mathbb{V}[y|x]$. This captures the well-known property of trade data that higher means are associated with higher variances. This assumption leads to equal weighting and Poisson distributed errors. In general, the following set of FOCs are solved to find $\hat{\beta}_{PPML}$: $\sum_{i} \left(y_i - \exp\left(x_i \hat{\beta}_{PPML}\right) \right) x_i = 0.$

¹³This is due to the simple fact that Jensen's inequality implies $\ln(\mathbb{E}[y]) \neq \mathbb{E}[\ln(y)]$.

logarithm of the number of new connections formed between countries j and k between t-1 and t:

$$s_{jkt}^{\text{core}} = ln(1 + (S_{jkt}^{\text{core}} - S_{jkt-1}^{\text{core}}))$$

$$\tag{3}$$

We motivate this measure as a proxy for information flows based on how syndicates are structured. In an underwriting syndicate, the lead underwriter holds primary responsibility for performing due diligence and shares information with the other syndicate members (non-lead underwriters). Therefore, information flows between the trading partners are strongest when the underwriter in the importing (issuing) or exporting country is the lead underwriter. β is the coefficient of interest, which we expect to be positive. In line with best practice for trade matrices with many zero entries, Equation (2) is estimated with PPML. For reference, we also provide results estimated with OLS.

While Equation (2) examines the contemporaneous relationship between newly formed core syndication ties and flows of underwriting services, we also estimate the same equation with several lags of the syndication variable to explore the persistence of the effects. A positive correlation between flows of underwriting services and past changes in the intensity of syndication ties is suggestive of causality. The causal interpretation of our baseline result, however, is potentially threatened by reverse causality. Indeed, underwriters from export markets might establish new linkages in import markets in anticipation of an increase in exports. We address the potential for reverse causality with a "placebo test" and an instrumental variable (IV) approach.

First, we perform an analysis akin to a placebo test. We test whether new syndication linkages formed by non-lead underwriters in the exporting and importing countries $(s_{jkt}^{\text{peripheral}})$ affect exports less strongly than new core syndication ties. As explained earlier, we expect new syndication ties between non-lead underwriters to be associated with a lower information content, so we should find a smaller effect on exports. Indeed, information acquisition is mainly conducted by lead underwriters. This analysis serves as a placebo test: if our benchmark result was purely driven by spurious correlation between syndication ties and exports of equity underwriting services, it would be equally likely to show up in the coefficient on the most informative and the least informative syndication ties. In addition, the test provides the first piece of evidence in support of our hypothesis that underwriting syndicates promote exports by mitigating information frictions. The empirical model is as follows:

$$X_{jkt} = \exp\left\{\beta_1 s_{jkt}^{\text{core}} + \beta_2 s_{jkt}^{\text{peripheral}} + \pi_{jt} + p_{kt} + \rho_{jk}\right\} + \varepsilon_{jkt}$$
(4)

All previously described variables are as before. $s_{jkt}^{\text{peripheral}}$ is defined in a similar manner to s_{jkt}^{core} in Equation (3) but using the stock of pheripheral linkages, $S_{jkt}^{\text{peripheral}}$, i.e. syndication linkages formed by non-lead underwriters in the exporting and importing countries. While we expect β_1 and β_2 to be positive, we expect β_2 to be smaller in magnitude or insignificant. In line with best practice for trade matrices with many zero entries, Equation (4) is estimated with PPML.

To further rule out reverse causality, we develop an instrumental variable (IV) approach. We aim to isolate the causal effect of syndicates on exports through information production (supply-side effects). Identification is endangered by the fact that underwriters from export markets might establish new linkages in import markets in anticipation of an increase in exports due to demand shocks in country k (demand-side effects). We identify supply-side shocks to new core syndication ties between exporting country j and importing country k by measuring shifts in country j's underwriters' interest in country k as an export destination unrelated to changes in demand-side prospects in country k. This IV approach follows the spirit of Calomiris et al. (2021). In particular, we instrument newly formed core ties between j and k with newly formed core ties between j and i. The subscript i denotes the importing country with the cumulative import value from j over the period 2000-2015 closest to that of country k. In other words, country i is comparable to country k as an export market for country j's underwriters. The idea is that newly formed core ties between j and i measure country i's attractiveness to country j's underwriters, which is plausibly exogenous to demand-side shocks to exports from i to k. Our relevance assumption is that positive shocks to the attractiveness of country i (newly formed core ties between j and i) constitute negative shocks to the attractiveness of country k. Our exogeneity assumption is that newly formed core syndication ties between j and i only affect exports from j to k through their impact on newly formed core ties between j and k. However, it is possible that syndication linkages between j and i may not be exogenous to exports from j to k if these exports are expected to be redirected to i. While we cannot rule out this possibility, it does suggest a channel by which syndicates matter for exports. The first stage is given by Equation (5).

$$s_{jkt}^{\text{core}} = \exp\left\{\beta s_{jit}^{\text{core}} + \pi_{jt} + p_{kt} + \rho_{jk}\right\} + \varepsilon_{jkt}$$
(5)

Due to the non-linear nature of our modeling procedure, we take a control function approach, whereby the estimated residuals from the first stage, $\hat{\varepsilon}_{jkt}$, are introduced as a control variable in the second stage to overcome endogeneity concerns. The second stage is given by Equation (6).

$$X_{jkt} = \exp\left\{\beta s_{jkt}^{\text{core}} + \gamma \hat{\varepsilon}_{jkt} + \pi_{jt} + p_{kt} + \rho_{jk}\right\} + \nu_{jkt}$$
(6)

Our final piece of analysis is to provide further evidence for the mechanism. If syndicates promote exports by mitigating information frictions as described in Section 2.4, the effect of new syndication ties will vary with the severity of information frictions between country j and country k. The main information frictions affecting trade in equity underwriting services relate to the barriers that investors face in accessing and interpreting price relevant information, particularly soft information, about foreign issuers. This problem is more acute when the issuing country is perceived as riskier. We first separate countries into OECD members and non-OECD members as a rough measure of contract enforcement and estimate Equation (2) for the four possible country-pair types. We then extend our analysis by adding a proxy for the riskiness of the importing (issuing) country, which we interact with our measure of syndication linkages. The estimating equations are:

$$X_{jkt} = \exp\left\{\beta_1 s_{jkt}^{\text{core}} + \beta_2 s_{jkt}^{\text{core}} \times \text{Risk}_{kt} + \pi_{jt} + p_{kt} + \rho_{jk}\right\} + \varepsilon_{jkt}$$
(7)

As before, π_{jt} are exporter-time fixed effects, p_{kt} are importer-time fixed effects, ρ_{jk} are country-pair fixed effects, and s_{jkt}^{core} is defined in Equation (3). Risk_{kt} is a proxy for risk associated with the importing (issuing) country.¹⁴ The information acquisition role of syndicates grows in importance when the importing country becomes riskier. Therefore, we expect the coefficient β_2 to be significantly positive when a higher value of Risk_{kt} indicates higher risk. The risk proxies we use are the destination country's score on the International Country Risk Guide index (ICRG index_{kt}), the destination country's sovereign credit rating as measured by Moody's (Credit rating_{kt}), the destination country's Resolving Insolvency score from the World Bank Doing Business surveys (Insolvency score_{kt}), and the destination country's Resolving Insolvency Recovery Rate score from the World Bank Doing Business surveys (Recovery score_{kt}). We expect that the effect of new core syndication linkages on exports is larger when the destination country scores worse on the ICRG index, has a worse sovereign credit rating, and performs worse in terms of resolving insolvencies. The proxies for risk are described in Section 4.4. Equation (7) is estimated with PPML.

Finally, we build on the insight that exports of underwriting services emanating from initial public offerings (IPOs) are likely to be more information sensitive than exports emanating from other transactions (follow-on offerings and convertible debt). Indeed, information asymmetries between issuers and investors are the largest in the case of IPOs, since the latter refer to the process of offering equity securities to the public in a new stock issuance for the first time. Follow-on offerings and convertible debt issuance refer to the issuance of equity securities or convertible debt following a company's IPO - at which stage market participants already have much more information on the issuer. Thus, we expect the effect of new core syndication ties to be higher for IPOs. We, therefore, estimate Equation (2) separately for exports of services related to IPOs and other transactions, expecting the coefficient β to be higher for the former. Such a strategy of splitting the sample according to the information sensitivity of transactions has been applied for example in Brei and von Peter (2018) when looking at international banking. The authors distinguish between interbank lending and credit extended to non-banks (corporates and non-bank financial institutions), the latter being more information sensitive.

4 Data sources and descriptive statistics

4.1 Exports of equity underwriting services

Our source of data on trade flows is Dealogic's Equity Capital Market (ECM) database. Our dependent variable is a measure of exports of equity underwriting services based on revenues earned by

¹⁴Note that there is no stand-alone Risk_{kt} control variable because it is absorbed by the importer-year FE.

banks from underwriting equity securities for their clients (issuers). Revenues are reported in current USD (nominal). Our data set covers 91,511 issues of equity securities offerings during the 2000-2015 period. These can be divided into 23,136 initial public offerings (IPOs), 58,454 follow-on offerings, and 9,921 convertible debt securities. We identify the key parties for each transaction – the underwriter(s) and the issuer. Every offering has as many revenue flows as there are underwriters involved. The 91,511 offerings in our data set constitute 306,512 revenue flows among underwriters and their clients. 57.3% of the offerings in our data set are underwritten by a single underwriter. The remaining transactions are underwritten by underwriting syndicates. When more than one underwriter is involved in an offering, revenue is apportioned by Dealogic's proprietary apportioning model among syndicate members. This is done based on their role in the syndicate (bookrunner, lead manager, co-manager, underwriter) and their level of participation in the deal, given by the fraction of newly issued securities allocated to them for selling to primary market investors.

In order to aggregate these individual revenue flows to a country-dyadic level, we need to identify the country of the underwriter(s) (exporting country/countries) and the country of the issuer (importing country) for each transaction. We adopt two conventions. First, the country of both the underwriter and the issuer is assigned based on their respective headquarters of operations, i.e. where their de-facto head office is located, rather than where each company is registered for tax or legal purposes. In other words, we are interested in de facto headquarters, where transactions are conducted, rather than de jure headquarters.¹⁵ Second, if the underwriter directly involved in a transaction is a subsidiary company, the exporting country is determined by the nationality of its parent company - in line with international trade statistics conventions (MSITS, 2010). Assigning the trade to the parent company also allows us to study commercial presence in the importing country as a mode of supply. We link the 91,511 offerings in our data set to 46,408 clients and 7,326 underwriter subsidiaries, controlled by 4,287 parent companies. Dealogic's ECM database contains information on the country of headquarters of operations for all the 46,408 issuers in our data set. The ECM database, however, does not include data on the location of underwriters. To fill this gap, we hand-collect data on addresses of headquarters of operations of underwriters' parent companies from corporate websites and databases, including Bureau van Dijk Orbis, Nexis UK and Bloomberg. We were able to collect data on countries of HQ of operations for 4,176 parent companies (97.4% of the total).

Data on revenue earned from underwriting equity securities in Dealogic's ECM database is either directly reported by underwriters or is estimated by Dealogic. Revenue data is directly disclosed for 45.1% of transactions by the underwriters involved in them, and undisclosed for the remaining 54.9% of transactions. However, Dealogic's ECM database offers proprietary estimates of revenue earned for 47.7% of transactions. We therefore have a combination of disclosed and estimated fees available for 92.8% of transactions. Using estimates is preferable to omitting transactions for which hard data on revenue earned is not available. While the use of estimates for a fraction of transac-

¹⁵This is to avoid any biases caused by companies registered or formally headquartered outside of the countries of their operations. For example, it is very popular for Chinese companies to register in the British Virgin Islands.

tions may be a source of random measurement error, the exclusion of transactions for which hard data is not available is likely to bias our results against countries that do not require underwriters to disclose revenue earned from equity underwriting for individual deals.¹⁶ Finally, 7.2% of transactions do not have disclosed or estimated revenue data available, and are therefore omitted from our sample. The missing geographical and revenue data means that we cannot observe 7,710 out of the 306,512 revenue flows (2.52%) in our data set. In terms of the pattern of missing data, we were less likely to find geographical data for very small and no longer operating underwriters with no trace-able online footprint. We would however not expect this to affect specific countries disproportionately.

After aggregating individual revenue flows into country-dyadic flows, we merge the latter with the CEPII distance data set. The country of origin is the country of the underwriter (exporting country), and the country of destination is the country of the issuer (importing country). Our country-dyadic flows derived from Dealogic data cover 122 countries of origin and 145 countries of destination. We replace any unmatched observations in the trade matrix with zeros. This is justified by the comprehensive nature of Dealogic's Equity Capital Markets database, which allows us to treat unobserved trade flows as true zeros with a very high level of confidence. Naturally, it cannot be ruled out that some of the zero trade flows in our data set may be the consequence of missing data, but we estimate that the proportion of "false zeros" is likely to be very small. All the missing data combined represent 7,710 revenue flows among underwriters and clients, which we are either unable to pinpoint geographically or for which we do not have data on revenue earned. We would expect that the vast majority of these missing revenue flows will be associated with country-dyads that have non-zero trade flows and only a small fraction of the 7,710 missing revenue flows is likely to fall on country-dyads for which we report zero trade flows. Therefore, the fraction of "false zeros" resulting from the missing data is likely to be very small. Even in the worst case scenario where all the missing revenue flows would fall exclusively on country-dyads for which we report zero trade flows, they would only create a small fraction of "false zeros" because the 7,710 missing revenue flows represent only 2.8% of all zero trade flows (271,205 trade flows in total). In other words, a maximum of 2.8% of our zero trade flows would be misreported as zeros instead of small non-zero trade flows. This means that our zero trade flows can be most accurately interpreted as a mix of mostly true zeros and some censored observations (at most 2.8% of the true zeros) that are very close to zero. For this reason we maintain that our zero trade flows are informative and should not be treated as missing data, but rather as true zeros. We

¹⁶To get a sense of the quality of Dealogic's revenue estimates, we contacted Dealogic and inquired about their estimation method. We learned that Dealogic employs a designated data analytics team, which focuses on modelling fees earned from investment banking transactions, including equity underwriting. Dealogic develops pricing models similar to those used by investment banks to price these transactions in the first place. The parameters of these models are not estimated from data on other transactions, but they are instead derived from regular expert interviews with industry practitioners responsible for pricing investment banking deals. We also inquired about the accuracy of such estimates and were advised that Dealogic's interviews with industry practitioners indicate that their estimates are within a 5% margin of error. Further reassurance on the reliability of the estimates comes from the fact that Dealogic serves as a platform for investment banks to collate their data on capital market transactions into a single data set that they rely on for market share calculations, ranking tables, and industry reports.

drop country pairs that do not display any variation in revenue flows (i.e. exports are always zero), as they are not informative for the estimation of the gravity equations. We obtain a sample of 132 countries that have at least one non-zero revenue flow during the 2000-2015 period (see list of countries in Appendix). This leaves us with a very sparse matrix of trade flows. The proportion of zero trade flows in our country-pair panel regressions is typically around 50% (see tables in Section 5 for exact numbers). However, the PPML estimation method is capable of handling such sparse trade matrices (Santos Silva and Tenreyro, 2006).

Our data set has several advantages over data sources used in the existing literature. Unlike data sets on financial services trade that rely on balance of payments statistics, our data does not require any harmonization, such as mirroring. It is recorded by a single data vendor, Dealogic, in a consistent manner across countries for every offering of equity securities. In contrast to other related research, which relies on the IMF's Coordinated Portfolio Investment Survey (CPIS) data, our dependent variable is built using revenue earned from the provision of financial services, rather than financial holdings. As explained above, this enables us to estimate a gravity equation that is consistent with a variety of models of international trade. Unlike CPIS data, the data used here does not omit transactions from any countries and the data collection process is globally consistent. In contrast, CPIS excludes some important countries, including China, and relies on surveys with varying levels of coverage of institutional investors across participating countries. As a result, the resultant dyadic observations of cross-border financial holdings captured by CPIS data are not necessarily measured consistently across countries and there are many missing values. By contrast, due to the comprehensive and consistent nature of the Dealogic ECM database, the zero trade flows in our data set can be interpreted as "real zeros" with a very high degree of confidence. Unlike missing values, they carry useful information about the structure of trade among countries.

4.2 Traditional proxies for bilateral trade costs

The traditional trade costs variables included in gravity models (distance, shared language, shared colonial history, and shared border) lend themselves perfectly to capture the bias toward more proximate or familiar markets (Sarkissian and Schill, 2004). We source these variables from the Centre d'Études Prospectives et d'Informations Internationales (CEPII) GeoDist distance data set (Mayer and Zignago, 2011). We begin with a bilateral population weighted distance variable. To account for the physical contiguity of countries, we include an indicator variable that equals one if countries share a political border and zero otherwise (*Shared border*). We also control for whether the trading partners share a language (*Shared language*) and whether they have a common colonial history (*Shared colonial history*). There are several variables available in the CEPII distance data set that capture the colonial history of countries. We opt for a non-directional specification. Specifically, *Shared colonial history* is equal to one for countries that were in a colonial relationship at any point in history, and is equal to zero otherwise. We do this to reflect the stylized fact that former colonies are not necessarily only importers, but can also be major exporters of financial services, the US being the most prominent

example.

4.3 Cross-border syndication ties

We measure cross-border syndication ties using deal-level data on offerings of equity securities from Dealogic's Equity Capital Market database between 2000 and 2015. The data set contains 91,511 equity securities offerings, out of which 39,075 (42.7%) are syndicated deals involving more than one underwriter. We follow a methodology similar to that of Caballero et al. (2018) for lending linkages to construct two novel measures of cross-border syndication ties. Caballero et al. (2018) construct a global network of banks, in which relationships are formed when banks extend syndicated loans to each other. They take into account the direction of the linkages, but not the amount lent, because they are interested in these lending flows only as a proxy for information flows. Specifically, they calculate the number of unique pairs of banks linked through syndicated loans extended by banks in the exporting country j to banks in the importing country k between the beginning of their sample period (1990) and year t. The authors then take a logarithmic transformation of the first difference in the number of unique pairs of banks.

In a similar fashion, we construct a global equity underwriter network, in which relationships are formed when underwriters are part of the same equity underwriting syndicate. We rely on equity offerings underwritten between 2000 and year t to construct a global underwriter network for year t. We assign each underwriter to a country using the headquarters of operations of the underwriter's parent, consistent with our geographical assignment of trade flows. We define the stock of syndication ties between countries j and k (S_{jkt}) as the number of unique pairs of equity securities underwriters connected through co-membership in underwriting syndicates between the year 2000 and year t. This is a count measure because we are interested in syndication ties as a proxy for information flows. Following Caballero et al. (2018), we then transform the aggregate number of syndication ties S_{ikt} by taking the natural logarithm of its first difference. This is because the question we ask is whether newly formed ties increase the flow of underwriting services between trading partners. In contrast to Caballero et al. (2018), who construct a directed measure of bank linkages from lenders to borrowers, our syndication ties are undirected and therefore $S_{jkt} = S_{kjt}$. Whereas information naturally flows from borrower (importer) to lender (exporter) in syndicated lending, there is no natural direction in which information flows in underwriting syndicates. Instead, information can flow in any direction among syndicate members.

That being said, the intensity of information flows among different types of syndicate members is unlikely to be constant. We exploit the structure of underwriting syndicates to identify the syndication relationships which are likely to generate the highest information flows. In an underwriting syndicate, the lead underwriter (bookrunner or lead manager) holds primary responsibility for performing due diligence and manages the relationship with the issuer on behalf of non-lead underwriters (co-managers or underwriters). Concurrently, non-lead underwriters, who are primarily tasked with selling equity securities, can feed back information from institutional investors to lead underwriters to inform the pricing of equity offerings. Information flows within underwriting syndicates are therefor expected to be strongest for underwriter-dyads that involve a lead underwriter. By extension, such linkages are expected to contribute more to overcoming information barriers among exporters and importers than those among non-lead underwriters. We denote syndication ties involving a lead underwriter as "core syndication ties". While information is still expected to flow between the trading partners when the underwriters in both countries are non-lead underwriters, the information content of these syndication ties should be lower. We denote syndication ties where the underwriters in both countries are non-lead underwriters as "peripheral syndication ties". Separating syndication ties into two categories based on the strength of the information flows they generate enables us to provide evidence for the information channel. Core syndication ties, S_{jkt}^{core} , are measured as the sum of unique pairs of underwriters for country-dyad jk connected through syndicate relationships where at least one of these underwriters is the lead underwriter. Peripheral syndication ties, $S_{jkt}^{\text{peripheral}}$, are measured by computing the sum of unique pairs of underwriters for country-dyad jk connected through syndicate relationships where both of the underwriters are in non-lead positions. For both core and peripheral ties, our explanatory variable is the natural logarithm of the number of new connections formed between countries j and k between t-1 and t:

$$s_{jkt}^{\rm h} = ln(1 + (S_{jkt}^{\rm h} - S_{jkt-1}^{\rm h})) \text{ where } h = \{\text{core, peripheral}\}$$
(8)

We also perform robustness tests using lags of s_{jkt}^{h} . This allows us to relax assumptions about the timeframe at which newly formed syndication ties influence exports in equity underwriting services.

4.4 Proxies for issuing country risk

4.4.1 ICRG Political Risk index

We use the International Country Risk Guide (ICRG) Political Risk index to proxy for country-specific risk. The ICRG index has been widely employed in the literature to proxy for contract enforcement and creditworthiness (see e.g., Berkowitz et al., 2006; Antràs and Foley, 201; Caballero et al., 2018). The index is published by the PRS Group and consists of 12 underlying sub-indices covering the following areas: (1) bureaucracy quality, (2) corruption, (3) democratic accountability, (4) external conflict, (5) ethnic tensions, (6) government stability, (7) internal conflict, (8) investment profile, (9) law and order, (10) military in politics, (11) religious tensions, (12) socioeconomic conditions. Each of these components is reported on a 1-100 scale, where 100 indicates the lowest risk, and 0 indicates the highest risk. A political risk rating of 0.0% to 49.9% indicates a very high risk; 50.0% to 59.9% high risk; 60.0% to 69.9% moderate risk; 70.0% to 79.9% low risk; and 80.0% or more very low risk (PRS Group, 2022). Since we do not have a prior on which of the 12 components of the ICRG index would be the best proxy for the quality of contract enforcement in a given country, we follow Caballero et al. (2018) and use the first principle component as our explanatory variable for measuring the risk of the destination (importing) country. The first principal component explains 49% of the variance and has

an eigenvalue of 5.9. The next component only explains an additional 11% and has an eigenvalue of 1.3. The first principal component loads positively on all the index components with loadings ranging from 0.17 on government stability, to 0.29 on investment profile and 0.34 on military in politics.

4.4.2 Moodys' sovereign credit ratings

Sovereign credit ratings have also been widely used in the literature to proxy for contract enforcement and creditworthiness (see e.g., Caballero et al., 2018). We use the destination country's sovereign credit rating as measured by Moody's as another proxy for country risk.¹⁷ Moody's credit ratings scale ranges from Aaa (highest rating) to C (lowest rating) and comprises 21 notches. We have transformed Moody's ratings into numerical categories, where a lower number indicates a higher credit rating and vice versa (from Aaa=1 to C=21). Ratings are divided into two categories, namely investment grade and speculative grade. Investment grade includes ratings Aaa - Baa3 and speculative grade includes ratings Ba1 - C. 96.1% of cross-border trade flows in equity underwriting services in our sample are with issuers headquartered in countries with an investment grade sovereign credit rating, while less than 4% of cross-border equity underwriting services are delivered to issuers in speculative grade countries. Credit ratings are not available for every year and country in our sample due to the coverage of countries being either initiated or suspended during our sample period. Overall, data on Moody's sovereign credit ratings is available for 86.5% of observations in the data set used for analysis (country-dyads that have at least one non-zero trade flow in the 2000-2015 period).

4.4.3 World Bank Doing Business - Resolving Insolvency

We use the destination country's scores on Resolving Insolvency from the World Bank Doing Business surveys for 2005-2015 (World Bank, 2022).¹⁸ The surveys collect data on the time, cost and outcome of insolvency proceedings involving domestic entities as well as the strength of the legal framework surrounding insolvency. The data are obtained from questionnaire responses by local insolvency practitioners and corroborated through a study of laws and regulations as well as public information on insolvency systems (World Bank, 2022). The surveys contain two measures that assess how well different countries resolve insolvency, namely the strength of insolvency framework index and the recovery rate. In addition, the data set contains an overall resolving insolvency score, which is an equally weighted average of the underlying scores on strength of insolvency framework and recovery rate. The strength of insolvency framework index is based on underlying scores on the commencement of proceedings, management of debtors' assets, reorganization proceedings and creditor participation. The recovery rate is defined as cents on the dollar recovered by secured creditors through reorganization, liquidation or debt enforcement (foreclosure or receivership) proceedings. In our analysis, we use the overall score on resolving insolvency as well as the sub-score for the recovery rate. The scores are measured on a scale from 0 to 100, where 0 represents the lowest and 100 represents the best perfor-

¹⁷We collected sovereign credit ratings from Moody's website. This data is available free of charge upon subscription and it covers 132 countries for the period 2000-2015.

¹⁸The World Bank Doing Business surveys are only available from 2005 onwards.

mance. The values of these scores inform how a country's performance compares to other countries. A positive change in these scores across time for a particular country suggests that it is closing the gap with the best performing country over time.¹⁹

4.5 Miscellaneous variables

4.5.1 Instrumental variable

As explained in Section 3, we instrument newly formed core ties between j and k with newly formed core ties between j and i, where i denotes an importing country with exports from j similar to those of country k. For every exporter - importer pair jk, we select an alternative importer i such that the value of exports from j to k is as closely as possible matched by the value of exports from j to i. Specifically, we use cumulative trade flows for 2000-2015 to choose the most similar alternative importer country i for each exporter-importer pair jk. Newly formed core syndication ties between jand i are then computed as described in Section 4.3.

4.5.2 Financial integration

We use two proxies for measuring financial integration. The first proxy is a measure of the stock of lending ties between the trading partners as in Caballero et al. (2018), constructed using data on interbank syndicated loans (Lending ties_{jkt}). The second proxy is a measure of the equity asset holdings of the exporting country (underwriter) in the importing country (issuer) as a percentage of the total overseas equity asset holdings of the exporting country (Equity holdings_{jkt}). This proxy is constructed using data from the IMF Coordinated Portfolio Investment Survey (CPIS).

To construct our *Lending ties* variable, we first obtain data on all (146,022) syndicated loans available in the Dealogic Loans Analytics database between January 1, 2000 and December 31, 2015. We then restrict the 'borrower type' to 'private sector bank' and 'public sector bank', leaving us with 8,019 syndicated loans. These loans are primarily underwritten as term loans (39.6% of loan tranches), fixed or floating certificates of deposit (19.9%), credit facilities (18.9%) and revolving credit (8.1%). Unfortunately, Dealogic does not offer a comparable categorisation for 'lender type'. This may result in some of the lenders not necessarily being banks. This is however unlikely to have a material impact on our results, given that despite this caveat, this measure remains a viable proxy for financial integration. Our data set of inter-bank syndicated loans covers 2,750 lenders from 110 countries and 1,647 borrowers from 135 countries. In order to identify lending ties among individual lenders and borrowers, we reshape the data set of syndicated loans such that each lender-borrower interaction is listed in a separate row, leading to a data set of 63,304 bank lending ties. We use data on the nationality of borrowers and lenders provided by Dealogic to link them to countries. We then construct a cumulative global bank network (GBN) for each year between 2000 and 2015 using all lending ties

¹⁹For example, a score of 75 in Doing Business 2019 means that an economy was 25 percentage points away from the best regulatory performance constructed across all economies and across time. A score of 80 in Doing Business 2020 would indicate that the economy was improving (World Bank, 2022).

between year 2000 and year t. Our proxy for financial integration is the number of bank pairs in countries j and k which are directly connected in the cumulative GBN in year t (AL_{jkt}) . Note that the direction of connection matters, given that AL_{jkt} is the number of banks in country j lending to banks in country k. Thus $AL_{jkt} \neq AL_{kjt}$.

We construct our *Equity holdings* variable using data on international equity securities holdings sourced from the Coordinated Portfolio Investment Survey (CPIS) released by the IMF (2022). CPIS reports data on foreign portfolio asset holdings, divided into equity, long-term debt and short-term debt, by residence of the issuer. We use the data on equity holdings, which covers 83 source countries and 214 destination countries. For the available country pairs, we obtain equity holdings held by investors from exporting country j (underwriter) in importing country k (issuer). We then calculate the percentage of the total overseas equity asset holdings of the exporting country invested in each destination country for every year between 2001 and 2015.²⁰ For example, the 2001 CPIS reports US foreign equity holdings for 114 destination countries. For 14 of these destination countries, it reports zero equity holdings by US investors. For those countries with non-zero US holdings, the percentage of US foreign equity holdings ranges from 0.000063% for Gambia to 21.993651% for the United Kingdom. In contrast, the 2015 CPIS reports foreign equity investments of US investors across as many as 209 countries, 116 of which have non-zero equity holdings ranging between 0.000015% for Burkina Faso and 13.876336% for the Cayman Islands. It is worth noting that CPIS data is subject to several limitations. Its coverage of countries is incomplete given that it requires voluntary participation on the part of the source countries which report their foreign investments. The number of countries covered has expanded over the years, increasing from 67 in 2001 to 81 in 2015. In addition to its limited coverage, CPIS data on foreign equity holdings is subject to additional limitations including under-reporting of assets due to incomplete institutional coverage of the survey, third-party holdings and problems in collection methods (Lane and Milesi-Ferretti, 2008). Despite these limitations, CPIS data on equity holdings offers a viable measure of financial integration at the country-pair level.

4.5.3 Ownership ties

Equity securities underwriters operate networks of subsidiaries across multiple countries. In some cases, an underwriter is a parent of a bank located in the issuer's country. If the issuer has a preexisting lending relationship with the subsidiary, it may seem natural for a parent to underwrite the issuer's equity offering. To control for this ownership mechanism, we build a proxy for parentsubsidiary networks (Ownership ties_{jkt}) using data on the parent-subsidiary ties among underwriters in Dealogic's ECM database. We establish 26,965 parent-subsidiary ties by identifying unique dyads of underwriter parents and subsidiaries involved in the underwriting of equity issues in 2000-2015. We complement these ties with data on the country of headquarters of operations for 6,469 subsidiaries

 $^{^{20}\}mathrm{CPIS}$ data is not currently available for the year 2000.

(88.3%), which we hand-collected from corporate websites, Bureau van Dijk Orbis, Nexis UK and Bloomberg. We discard ties for which we do not have data on the country of headquarters of operations of the subsidiaries, leaving us with 25,229 ties (93.6% of initial total). The Dealogic ECM database, however, continuously updates the parent-subsidiary ties with the most recent ownership data available. This means that while the records on underwriter subsidiaries for each transaction are historically accurate, the corresponding parent companies are overwritten with the current, rather than historical, ownership of said subsidiaries. To allow our connectivity measure to vary over time, we correct for any changes in these subsidiary-parent ties between 2000 and 2015 using records of mergers and acquisitions from Dealogic's M&As database and from Nexis UK. Specifically, we corrected the records of parent companies linked to subsidiaries on an annual basis for 2,256 ties (8.9% of the total). This leads to a data set of time-varying parent-subsidiary ties recorded annually, which features 25,229 ownership ties among 6,469 underwriter subsidiaries controlled by 3,741 parent companies for the 2000-2015 period.

Using this corrected data, we construct a count measure for each country-dyad, which is equal to the number of underwriters that have subsidiaries in both countries. We therefore obtain an undirected measure of underwriter overlaps measured at the level of country-dyads.²¹ Our data set contains 800 country-dyads (3.2% of all dyads)²² that are connected by underwriter overlaps in at least one year of the sample period. The number of overlapping underwriters ranges from 1 to 30 and is time-varying. Finally, we normalize the count variable by dividing it by the sum of ownership ties for each exporter-year (i.e. the number of ties between the exporting country and the rest of the world). We refer to this variable as *ownership ties*.

5 Effect of syndication ties on exports

5.1 Baseline results

Table 1 below reports the results from the estimation of Equations (1) and (2).²³ Column (1) is Equation (1) estimated with PPML, column (2) is Equation (2) estimated with PPML. Column (3) is Equation (1) estimated with OLS, and column (4) is Equation (2) estimated with OLS. The Table reports the number of exporting and importing countries, as well as the proportion of zero trade flows in the sample for the PPML estimations. By construction, OLS drops the zero trade flows and in that

²¹For example, if say JPMorgan (parent) has subsidiaries in both the US and the UK, that counts as one ownership tie, regardless of how many subsidiaries JPMorgan has in each country. If for example JPMorgan, Morgan Stanley and Goldman Sachs have some subsidiaries in the US and the UK, regardless of their number, this would be three ownership ties between the US and the UK. Every parent company counts only once at most.

²²Given that our specification of ownership ties is undirected, we only count each country-dyad once here.

²³The coefficients are interpreted as follows. For dummy variables, changing the dummy from zero to one changes predicted exports by $(exp(\beta) - 1) \times 100\%$, where β is the regression coefficient. For variables expressed as natural logarithms, the coefficient β is an elasticity. In other words, a 1% increase in the explanatory variable is associated with a $\beta\%$ change in exports.

sense only takes into account the intensive margin of trade.

	(1)	(2)	(3)	(4)
$1 (\mathbf{D}^{*} + \cdots + \mathbf{n})$	0 5 40***			
$\ln(\mathbf{Distance}_{jk})$	$-0.540^{-0.01}$		$-0.500^{-0.100}$	
	(0.122)		(0.108)	
Shared language _{sh}	0.262		0.433*	
	(0.394)		(0.241)	
Shared $border_{jk}$	0.118		0.778^{**}	
	(0.452)		(0.352)	
Shared colonial history	1 691***		1 165***	
Shared colonial motor y_{jk}	(0.265)		(0.134)	
S_{jkt}^{core}		0.243^{***}		0.379^{***}
		(0.0315)		(0.0273)
Constant	18 89***	10 19***	14 88***	14 62***
Constant	(0.396)	(0.122)	(0.179)	(0.0269)
	(0.000)	(0.122)	(0.110)	(0.0205)
Observations	76,766	14,997	$6,\!875$	6,039
Exporter-year FE (jt)	Yes	Yes	Yes	Yes
Importer-year FE (kt)	Yes	Yes	Yes	Yes
Pair FE (jk)	No	Yes	No	Yes
Number of exporters	96	67	63	47
Number of importers	108	97	92	72
Prop. of zero trade flows	0.902	0.539	0	0
R^2	0.935	0.986	0.536	0.837

Table 1: The contemporaneous effect of core syndication ties

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01.

Standard errors clustered on origin and destination countries.

For PPML regressions the pseudo R^2 is reported, for OLS regressions the adjusted R^2 is reported.

Column (1) reports the PPML estimates from Equation (1). Regarding distance, we find that it is highly significant, but the distance elasticity is modest. A coefficient of -0.540 implies that a 10% increase in bilateral distance translates into a 5.4% decrease in exports of equity underwriting services. It is not surprising to find a small distance elasticity given the non-physical nature of trade in financial services. This is in sharp contrast to the literature on trade in goods where the distance elasticity varies between -0.9 and -1.1 (Head and Mayer, 2014). Our estimates are quantitatively in line with those of Portes and Rey (2005) for equities. Nevertheless, it might appear puzzling that physical distance still matters at all. The consensus in the literature on gravity in international finance is that

distance reflects information frictions. For example, Brei and von Peter (2018) find that the role of distance remains substantial for banking where transport cost is immaterial, pointing to the role of information frictions as a common driver. A shared language does not exhibit any significance. This is in contrast to the fact that language has been found to be an important determinant in services trade (see e.g., Ceglowski, 2006; Kimura and Lee, 2006; Walsh, 2008). A shared border is also insignificant. This result is in contrast to the literature on trade in goods where a shared border is usually highly significant, with an average coefficient of 0.7 (Head and Mayer, 2014). It is not surprising, however, since transportation costs associated with the provision of equity underwriting services are very small and many international financial centers are located in countries that do not share a physical border with each other. Finally, a shared colonial past has a significantly positive impact. A coefficient of 1.691 indicates that being a previous colonial partner increases exports by 442.5%. This coefficient is higher than magnitudes previously found in the goods trade literature (0.75) (Head and Mayer, 2014). The apparent importance of a shared colonial history is intuitive. Countries that have been in a colonial relationship are more likely to share common institutional and legal frameworks, which are paramount for trade in financial services. For reference, column (3) reports the OLS estimates from Equation (1). OLS drops all the zero trade flows, so effectively only looks at the intensive margin of trade. The distance elasticity is almost unaffected, but the effect of a shared colonial past decreases to 220.6%. A shared language now attracts significance at the 10% level and a shared border at the 5%level. Given the importance of zero trade flows in the sample, and the information that these flows carry, we use PPML as our baseline.

Column (2) reports the PPML estimates from Equation (2) with the change in core syndication linkages as the variable of interest. The traditional time-invariant bilateral gravity variables are absorbed by the country-pair fixed effects. Newly formed core syndication ties have a significantly positive impact on exports of equity underwriting services. Specifically, a 1% increase in new core syndication ties increases exports by 0.243%. In other words, doubling the change in the intensity of core ties is associated with a 24.3% increase in exports. For reference, column (4) reports the OLS estimates for the same equation. The effect of doubling the change in the intensity of core ties increases to 37.9%. This suggests that using OLS (and dropping the zero trade flows) would lead to an overestimation of the effect of syndication ties. We therefore use PPML as our baseline.

In Appendix Table A1, we perform a couple of robustness checks for the results in column (2). First, we control for the degree of financial integration between the trading partners. Second, we argue that ownership ties do not drive our results. There are cases where the underwriter is a parent of a bank located in the issuer country. If the issuer is already banking with the subsidiary, it would be natural for the parent to underwrite the issue. Despite its apparent plausibility, we argue that this mechanism is very unlikely to explain our results. In addition, we explicitly control for parent-subsidiary ownership ties and show that our results still hold.

5.2 Exploring persistence

We have so far uncovered a positive contemporaneous correlation between changes in syndication ties between two countries and flows of equity underwriting services between those two countries. In this section, we examine whether the effect of syndication ties is persistent, i.e whether lagged changes in syndication ties are positively correlated with flows of equity underwriting services. Persistence would be suggestive of supply-driven causality. This is because increases in syndication ties are equivalent to increased inflows of equity capital (whether from domestic or foreign investors) which put upward pressure on stock prices, reduces the cost of equity capital, and therefore encourages destination countries' firms to issue equity, a decision that occurs with a lag. A finding of persistence should alleviate concerns of reverse causation. However, it would only be indicative of supply-driven forces because serial correlation in demand shocks could produce a relation between changes in syndication ties (lagged inflows) and current issuances (current equity underwriting services). In the next sections, we will provide more evidence of a supply-driven mechanism by conducting a placebo test, and providing evidence to support that syndication ties mitigate information frictions.

We estimate Equation (2) with several lags of newly formed core syndication ties. The results of PPML estimations are in Table 2. We only report the results for the first two lags as the coefficients on the second and further lags are insignificant. Newly formed core syndication ties display a significantly positive correlation with exports of equity underwriting services at the first lag. In other words, core ties formed between t - 2 and t - 1 affect trade at time t. The contemporaneous effect is the largest. A coefficient of 0.0830 on the first lag in column (2) implies that a 1% increase in new core syndication ties increases exports by 0.0830%. In other words, doubling the change in the intensity of core ties between t - 2 and t - 1 is associated with a 8.3% increase in exports. In addition, the coefficient on ties formed between t - 1 and t is only slightly affected by the inclusion of the first lag. The effect on exports of doubling the change in the intensity of core ties between t - 1 and t decreases from 24.3% to 23% when the first lag is included.

	(1)	(2)	(3)
$s^{f core}_{jkt}$	0.243^{***}	0.230^{***}	0.234^{***}
	(0.0315)	(0.0326)	(0.0344)
$s^{\mathbf{core}}_{jkt-1}$		0.0830^{***}	0.0848^{***}
		(0.0208)	(0.0210)
$s^{f core}_{jkt-2}$			0.0339
-			(0.0227)
Constant	19.12***	18.85^{***}	18.72^{***}
	(0.122)	(0.168)	(0.222)
Observations	$14,\!997$	$14,\!150$	$13,\!156$
Exporter-year FE (jt)	Yes	Yes	Yes
Importer-year FE (kt)	Yes	Yes	Yes
Pair FE (jk)	Yes	Yes	Yes
Number of exporters	67	67	67
Number of importers	97	97	96
Prop. of zero trade flows	0.539	0.533	0.526
Pseudo R^2	0.986	0.986	0.986

 Table 2: The lagged effect of cross-border syndication ties

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01Standard errors clustered on origin and destination countries.

5.3 Addressing reverse causality

The main concern with our identification strategy is reverse causality. Reverse causality could drive the results if underwriters operating in export markets anticipate an increase in exports and therefore choose to form syndication ties with underwriters in import markets. Changes in syndication ties could be entirely demand-driven. Demand-side factors are any changes in the importing (issuing) country that affect investment opportunities, such as changes in productivity, technology, or local economic conditions. This is a particular concern since syndicates are groups of underwriters organized for the purpose of bringing an issue of equity securities to the market (which generates the exports) and dissolved thereafter. In other words, our measure of syndication linkages has a direct connection to trade activity.

We test for reverse causality by regressing new core ties (s_{jkt}^{core}) on the logarithm of contemporaneous and lagged exports $(ln(X_{jit}) \text{ and } ln(X_{jit-1}))$. The estimation is performed with OLS and a full set of fixed effects.²⁴ We find that the effect of contemporaneous exports on new core syndication ties is significantly positive while that of lagged exports is insignificant. However, the effect of exports is

²⁴The table of results is available upon request.

small in magnitude. A 1% increase in exports increases the change in the intensity of core ties by 0.08%. This means that the elasticity of new core syndication ties with respect to exports is roughly three times smaller than the elasticity of exports with respect to core syndication ties formed between t-1 and t. Nevertheless, this indicates that reverse causality is a priori a valid concern. Therefore, to provide empirical evidence that reverse causality cannot explain all our results, we conduct two additional tests. First, we conduct a kind of placebo test which relies on the intensity of information production associated with different syndication ties. Second, we use the IV strategy described earlier.

5.3.1 Information content of syndication ties

We test whether new syndication linkages formed by non-lead underwriters in the exporting and importing countries $(s_{jkt}^{\text{peripheral}})$ affect exports less strongly than syndication ties where the underwriter in either the importing or exporting country is the lead underwriter (s_{jkt}^{core}) . We expect new syndication ties between non-lead underwriters to be associated with a lower information content because information acquisition is mainly conducted by lead underwriters. Hence, we should find a smaller effect on exports. This analysis serves as a placebo test: if our benchmark result was purely driven by spurious correlation between syndication ties and exports of equity underwriting services, it would be as likely to show up in the coefficient on the most informative and the least informative syndication ties. In addition, the test provides the first piece of evidence in support of our hypothesis that underwriting syndicates promote exports through information production.

Table 3 presents the PPML results. Column (1) is our baseline from column (2) in Table 1. Column (2) is the estimation of Equation (2) with $s_{jkt}^{\text{peripheral}}$ as the explanatory variable instead of s_{jkt}^{core} . Finally, column (3) presents the results from the estimation of Equation (4) with both explanatory variables. The results are in line with our hypothesis and the channel we propose. In column (2), a 1% increase in new peripheral syndication ties increases exports by 0.0778%. In other words, doubling the change in the intensity of peripheral ties is associated with a 7.8% increase in exports. This effect is roughly three times smaller than that of new core syndication ties in column (1). When controlling for both syndication measures in column (3), the effect of doubling the change in the intensity of core ties is stable (at 23.2%), whereas that of peripheral ties falls dramatically to 3.3%. In this case, the effect of a 1% increase in new core syndication ties on exports is roughly seven times larger than the effect of new peripheral ties.

	(1)	(2)	(3)
$s_{jkt}^{\mathbf{core}}$	$\begin{array}{c} 0.243^{***} \\ (0.0315) \end{array}$		$\begin{array}{c} 0.232^{***} \\ (0.0327) \end{array}$
$s_{jkt}^{\mathbf{peripheral}}$		$\begin{array}{c} 0.0778^{***} \\ (0.0164) \end{array}$	0.0330^{**} (0.0160)
Constant	$19.12^{***} \\ (0.122)$	$19.74^{***} \\ (0.0665)$	$19.03^{***} \\ (0.126)$
Observations	14,997	14,997	14,997
Exporter-year FE (jt)	Yes	Yes	Yes
Importer-year FE (kt)	Yes	Yes	Yes
Pair FE (jk)	Yes	Yes	Yes
Number of exporters	67	67	67
Number of importers	97	97	97
Prop. of zero trade flows	0.539	0.539	0.539
$\mathbf{Pseudo} \ R^2$	0.986	0.985	0.986

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01Standard errors clustered on origin and destination countries.

5.3.2 Instrumental variables estimation

The results of our instrumental variables estimation are presented in Table 4. Column (1) reports the results of the first-stage estimation (Equation (5)), while column (2) shows the second-stage estimation (Equation (6)). The first stage is strong with an F-statistic of 13.54. The coefficient on s_{jit}^{core} is negative and significant at the 1% level - indicating that new core syndication ties between jand i are negatively correlated with those between j and k. The second-stage results provide further support for our baseline finding. We find that a 1% increase in new core syndication ties increases exports by 0.221%. In other words, doubling the change in the intensity of core ties is associated with a 22.1% increase in exports. While we cannot fully rule out endogeneity concerns, our tests taken together place limits on the extent to which reverse causality might be driving our results.

	First stage	Second stage
$S_{jit}^{\mathbf{core}}$	-0.0696^{***} (0.0189)	
$s_{jkt}^{\mathbf{core}}$		0.221^{***} (0.0656)
Response residual		0.0500 (0.0642)
Constant	$\begin{array}{c} 0.0615^{***} \\ (0.0155) \end{array}$	18.00^{***} (0.156)
Observations Exporter-year FE (jt)	14,068 Yes	9,029 Yes
Importer-year FE (kt) Pair FE (jk)	Yes Yes	Yes Yes
Number of exporters	72 72	52 66
Prop. of zero trade flows Pseudo R^2	0.642	$0.446 \\ 0.965$

Table 4: IV estimation

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01Standard errors clustered on origin and destination countries.

5.4 Assessing the information mechanism

In this section, we test whether the mechanism through which syndication linkages affect exports is related to information frictions. Using two different approaches, we find evidence supporting this hypothesis. First, we demonstrate that the effect of syndication linkages is larger in magnitude when the issuing country is riskier. To do so, we first separate countries into OECD and non-OECD as a rough measure of contract enforcement and estimate Equation (2) for the four possible country-pair types. We then estimate Equation (7) using several proxies for destination country risk described in Section 4.4. Second, we show that the effect of syndication linkages is larger for exports that relate to initial public offerings (IPOs) as opposed to other transactions (follow-on offerings and convertible debt) - the former being more information sensitive.

5.4.1 Do syndication ties matter more when the issuing country is riskier?

First, we follow the simple approach of splitting importers and exporters into groups by OECD membership, where the latter serves as a proxy for contract enforcement or more generally rule of law (Hall and Jones, 1999; Rodrik et al., 2004). While this is a crude proxy for destination country risk, OECD membership has the advantage of being completely predetermined and unaffected by subsequent changes in syndication ties or trade. It is therefore often used as a proxy for contract enforcement (e.g. Caballero et al., 2018). The results are presented in Table 5. We find that new core syndication ties are more important when the exporter is an OECD member and the importer is not (column (2)) than when both countries are OECD members (column (1)). The effect of doubling the change in the intensity of core ties is 5.6 percentage points larger when the destination country is not an OECD member. The results of column (3) have to be taken with extreme caution because the standard error is very wide due to the low number of observations. Indeed, there are very few transactions where the exporting country is not an OECD member, but the destination country is. Column (4), where neither country is an OECD member, also suffers from a lower number of observations but the coefficient is significant, but lower than in columns (1) and (2). These results are intuitive. Underwriters put their reputation and money at stake when they underwrite equity offerings. This means that we would expect the *difference* in contract enforcement, or more generally risk, between the exporting and importing countries to affect the value of syndication ties in mitigating this risk. Crucially, syndication ties are most needed when the importing country is riskier than the exporting country, but not necessarily vice versa or when both countries have a similar risk profile.

	()	(-)	(-)	()
	(1)	(2)	(3)	(4)
OECD membership	Both	Only origin	Only destination	Neither
$s_{jkt}^{\mathbf{core}}$	0.228***	0.284^{***}	0.258	0.165***
	(0.0290)	(0.0615)	(0.246)	(0.0415)
Constant	19.46***	17.48***	15.87***	19.10***
	(0.121)	(0.105)	(0.260)	(0.160)
Observations	6,020	4,114	909	1,929
Exporter-year FE (jt)	Yes	Yes	Yes	Yes
Importer-year FE (kt)	Yes	Yes	Yes	Yes
Pair FE (jk)	Yes	Yes	Yes	Yes
Number of exporters	29	24	18	31
Number of importers	32	59	20	38
Prop. of zero trade flows	0.44	0.511	0.592	0.484
Pseudo R^2	0.989	0.958	0.947	0.995

Table 5: OECD membership

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01Standard errors clustered on origin and destination countries.

Second, we estimate Equation (7) using several proxies for destination country risk described in Section 4.4. The results are in Table 6. The results of column (1) suggest that a 1% increase in new core syndication ties increases exports by 14.2% more when the destination country is at the 25th percentile of the ICGR index distribution as opposed to the 75th percentile. Since a higher value of the index

indicates lower destination country risk, this means that new core syndication ties matter more when the destination country is riskier.²⁵ The results of column (2) suggest that a 1% increase in new core syndication ties increases exports by 18.3% more when the destination country is at the 75th percentile of the Moody's credit rating distribution as opposed to the 25th percentile. Since a higher value of the rating indicates higher destination country risk, this means that new core syndication ties matter more when the destination country is riskier. The results of column (3) suggest that a 1% increase in new core syndication ties increases exports by 24.6% more when the destination country is at the 25th percentile of the distribution of scores on Resolving Insolvency as opposed to the 75th percentile. Since a higher value of the score indicates lower destination country risk, this means that new core syndication ties matter more when the destination country is riskier. Finally, the results of column (4) suggest that a 1% increase in new core syndication ties increases exports by 14.2% more when the destination country is at the 25th percentile of the distribution of scores on the recovery rate as opposed to the 75th percentile. Since a higher value of the score indicates lower destination country risk, this means that new core syndication ties matter more when the destination country risk, this means that new core syndication ties matter more when the destination country risk, this means that new core syndication ties matter more when the destination country risk, this means that new core syndication ties matter more when the destination country is riskier.

²⁵The differential impact of the explanatory variable s_{jkt}^{core} on exports for a country with Risk_{kt} at the 25th percentile of the distribution (Risk_{kt}^{25}) and a country with Risk_{kt} at the 75th percentile of the distribution (Risk_{kt}^{75}) is given by $exp(\beta_1) \left(exp(\beta_2 \times \text{Risk}_{kt}^{25}) - exp(\beta_2 \times \text{Risk}_{kt}^{75})\right) \times 100\%$, where β_1 is the coefficient on s_{jkt}^{core} and β_2 is the coefficient on $s_{jkt}^{\text{core}} \times \text{Risk}_{kt}$.

	(1)	(2)	(3)	(4)
$s_{jkt}^{\mathbf{core}}$	$\begin{array}{c} 0.331^{***} \\ (0.0549) \end{array}$	$\begin{array}{c} 0.204^{***} \\ (0.0396) \end{array}$	0.634^{***} (0.118)	0.440^{***} (0.0932)
$s_{jkt}^{\mathbf{core}} \times \mathbf{ICRG} \ \mathbf{index}_{kt}$	-0.0354^{**} (0.0157)			
$s_{jkt}^{\mathbf{core}} \times \mathbf{Credit} \ \mathbf{rating}_{kt}$		0.0148^{**} (0.00582)		
$s_{jkt}^{\mathbf{core}} \times \mathbf{Insolvency} \ \mathbf{score}_{kt}$			$\begin{array}{c} -0.00529^{***} \\ (0.00156) \end{array}$	
$s_{jkt}^{\mathbf{core}} \times \mathbf{Recovery} \ \mathbf{score}_{kt}$				-0.00246** (0.00112)
Constant	$19.15^{***} \\ (0.120)$	$\begin{array}{c} 19.23^{***} \\ (0.135) \end{array}$	$ \begin{array}{r} 18.38^{***} \\ (0.124) \end{array} $	$ \begin{array}{c} 18.33^{***} \\ (0.118) \end{array} $
Observations	$14,\!572$	13,687	9,545	9,545
Exporter-year FE (jt)	Yes	Yes	Yes	Yes
Importer-year FE (kt)	Yes	Yes	Yes	Yes
Pair FE (jk)	Yes	Yes	Yes	Yes
Number of exporters	67	66	64	64
Number of importers	88	84	93	93
Prop. of zero trade flows	0.539	0.538	0.508	0.508
Pseudo R^2	0.986	0.987	0.975	0.975

Table 6: Syndication ties and destination country risk

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01Standard errors clustered on origin and destination countries.

5.4.2 Do syndication ties matter more for more information sensitive services?

Finally, we show that the effect of new core ties is larger for exports that relate to IPOs rather than other issuance activities (follow-on offerings and convertible debt) - the former being more information sensitive. We estimate Equation (2) for four samples, namely all transactions, IPOs, follow-on offerings (FOs) and convertible debt, and FOs only. The results are in Table 7. Column (1) is our baseline for all types of transactions from column (2) in Table 1. The coefficient in column (2) suggests that doubling the change in the intensity of core ties is associated with a 28% increase in exports of equity underwriting services related to IPOs. This effect is 3.7pp larger than the effect for all transactions in column (1), 6.5pp larger than the effect for FOs and convertible debt together (column (3)), and 5pp larger than the effect for FOs only (column (4)).

	(1)	(2)	(3)	(4)
	All	IPOs	FOs and convertibles	FOs
s_{jkt}^{core}	0.243***	0.280***	0.215***	0.230***
	(0.0315)	(0.0373)	(0.0294)	(0.0333)
Constant	19.12***	18.10***	18.87***	18.60***
	(0.122)	(0.150)	(0.112)	(0.124)
Observations	$14,\!997$	9,824	$11,\!678$	$11,\!185$
Exporter-year FE (jt)	Yes	Yes	Yes	Yes
Importer-year FE (kt)	Yes	Yes	Yes	Yes
Pair FE (jk)	Yes	Yes	Yes	Yes
Number of exporters	67	58	54	54
Number of importers	97	87	83	82
Prop. of zero trade flows	0.539	0.419	0.458	0.443
Pseudo R^2	0.986	0.981	0.981	0.979

Table 7: IPOs versus FOs and convertible debt issuance

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01

Standard errors clustered on origin and destination countries.

All the above results are consistent with the mechanism we propose: syndication linkages contribute to trade growth by mitigating risk through information acquisition. While they do not prove that the effect of syndication ties on trade is causal, one would be hard pressed to find an alternative story that would be consistent with all the evidence presented.

6 Conclusions

We shed light on the relationship between cross-border syndication ties and trade in equity underwriting services using theory-consistent gravity equations. The interest in gravity in international finance is not new, but theory-consistent gravity estimations for trade in financial services are extremely rare. This scarcity is undoubtedly related to the limitations of current data on financial services. Our first contribution is therefore to construct a novel panel data set on revenue earned on domestic and crossborder equity securities underwriting transactions using the Dealogic Equity Capital Market database. Our data set covers 91,511 deals across 122 countries of origin and 145 countries of destination for the period 2000-2015. We argue that our data set has several advantages over other data sets based on balance of payments statistics. Most importantly, having access to transaction-level data enables us to build novel measures of core and peripheral syndication ties based on underwriters' co-membership of, and role in, underwriting syndicates. In the empirical implementation of our model, we account properly for multilateral resistance with country-time fixed effects, country-pair fixed effects, and apply state-of-the-art econometric methods (PPML). We find that newly formed core syndication ties are associated with a statistically and economically significant increase in exports of equity underwriting services.

We argue that international syndication ties mitigate information frictions, which the literature argues are a major impediment to trade in financial services. We support this conjecture with several pieces of evidence. First, we show that newly formed peripheral syndication ties have a smaller positive effect on exports than core syndication ties. Peripheral syndication ties are associated with lower information flows between trading partners. The results therefore support the conjecture that syndicates have an impact on trade through the mitigation of information frictions. Second, we demonstrate that new core syndication ties have a stronger effect on exports of equity underwriting services when information frictions between the trading partners are more severe. We do this in two ways. First, we exploit cross-sectional variation in the risking so of the issuing countries. The main information frictions affecting trade in equity underwriting services relate to the barriers that investors face in accessing and interpreting price relevant information about foreign issuers. This problem is more acute when the issuing country is perceived as riskier. We find that new core syndication ties have a stronger effect on exports of equity underwriting services when the issuing country is riskier. Second, we exploit cross-sectional variation in the information sensitivity of transactions. Exports of underwriting services emanating from primary equity issuance activity (IPOs) are likely to be more information sensitive than exports emanating from other transactions (follow-on offerings and convertible debt). Accordingly, we find that the effect of new core syndication ties is higher for IPOs.

Reverse causality is a priori a concern for the causal interpretation of our results. However, we provide several pieces of analysis that place limits on the extent to which reverse causality might be driving our findings. These include evidence for the fact that core syndication ties matter more than peripheral ones due to their higher information production, and IV estimation results that bolster our baseline findings. While we cannot fully rule out endogeneity concerns, our paper provides strong evidence for the effect that new syndication ties have on exports of equity underwriting services through the mitigation of information frictions.

References

Ahearne, A.G., Griever, W.L., and Warnock, F.E., 2004. Information costs and home bias: an analysis of US holdings of foreign equities. Journal of International Economics 62(2), 313-336.

Aichele, R., Felbermayr, G. J. and Heiland, I., 2014. Going Deep: the Trade and Welfare Effects of TTIP, Munich, CESifo, Working Paper No. 5150.

Anderson, J.E., Borchert, I., Mattoo, A., and Yotov, Y.V., 2018. Dark costs, missing data: Shedding some light on services trade. European Economic Review, 105(C), 193-214.

Antràs, P., Foley, F.C., 2015. Poultry in motion: a study of international trade finance practices. The Journal of Political Economy, 123(4), 853-901.

Arkolakis, C., Costinot, A., and Rodriguez-Clare, A., 2012. New trade models, same old gains?. American Economic Review 102(1), 94-130.

Baldwin, R., and Taglioni, D., 2006. Gravity for dummies and dummies for gravity equations. National Bureau of Economic Research Working Paper 12516.

Berger, E.A., Butler, A.W., Hu, E., and Zekhnini, M., 2021. Financial integration and credit democratization: Linking banking deregulation to economic growth. Journal of Financial Intermediation 45, 100857.

Berkel, B., 2007. Institutional determinants of international equity portfolios - a country level analysis. The B.E. Journal of Macroeconomics 7(1), 1-33.

Berkowitz, D., Moenius, J., Pistor, K., 2006. Trade, law, and product complexity. The Review of Economics and Statistics, 88(2), 363-373.

Bharath, S., Dahiya, S., Saunders, A., and Srinivasan, A., 2007. So what do I get? The bank's view of lending relationships. Journal of Financial Economics, 85(2), 368-419.

Brei, M., and von Peter, G., 2018. The distance effect in banking and trade. Journal of International Money and Finance, 81(C), 116-137.

Caballero, J., Candelaria, C., and Hale, G., 2018. Bank linkages and international trade. Journal of International Economics, 115, 30-47.

Calomiris, C.W., Larrain, M., and Schmukler, S.L., 2021. Capital inflows, equity issuance activity, and corporate investment. Journal of Financial Intermediation, 46, 100845.

Ceglowski, J., 2006. Does Gravity Matter in a Service Economy? Review of World Economics / Weltwirtschaftliches Archiv, 142(2), 307-329.

Chan, K., Covrig, V., and Ng, L., 2005. What determines domestic bias and foreign bias? Evidence from mutual fund equity allocations worldwide. Journal of Finance 60(3), 1495-1534.

Chiţu, L., Eichengreen, B., and Mehl, A., 2014. History, gravity and international finance. Journal of

International Money and Finance 46, 104-129.

Chuluun, T., 2015. The role of underwriter peer networks in IPOs. Journal of Banking and Finance 51, 62-78.

Claessens, S., and van Horen, N., 2021. Foreign banks and trade. Journal of Financial Intermediation, 45, 100856, 1-21.

Coeurdacier, N., and Martin, P., 2009. The geography of asset trade and the euro: insiders and outsiders. Journal of Japanese and International Economies 23(2), 90-113.

Coeurdacier, N., and Guibaud, S., 2011. International portfolio diversification is better than you think. Journal of International Money and Finance 30(2), 289-308.

Contessi, S., de Nicola, F., 2012. What do we Know about the Relationship Between Access to Finance and International Trade? Federal Reserve Bank of St. Louis WP 2012-054.

Corwin, S. A., and Schultz, P., 2005. The Role of IPO Underwriting Syndicates: Pricing, Information Production, and Underwriter Competition. The Journal of Finance 60(1), 443-486.

Coval, J., and Moskowitz, T., 1999. Home Bias at Home: Local Equity Preference in Domestic Portfolios. Journal of Finance 54, 2045-2073.

Coval, J., and Moskowitz, T., 2001. The Geography of Investment: Informed Trading and Asset Prices. Journal of Political Economy 109, 811-841.

Daude, C., and Fratzscher, M., 2008. The pecking order of cross-border investment. Journal of International Economics 74(1), 94-119.

Dunbar, C.G., 2000. Factors affecting investment bank initial public offering market share. Journal of Financial Economics 55, 3–41.

Faruqee, H., Li, S., Yan, I.K., 2004. The determinants of international portfolio holdings and home bias. IMF Working Paper 2004-34.

Forbes, K.J., 2010. Why do foreigners invest in the United States? Journal of International Economics 80(1), 3–21.

Francois, J., and Hoekman, B., 2010. Services Trade and Policy. Journal of Economic Literature, 48, 642–692.

Francois, J., and Wignaraja, G., 2008. Economic Implications of Asian Integration. Global Economy Journal, 8(3).

French, K., and Poterba, J., 1991. Investor Diversification and International Equity Markets. American Economic Review 81, 222-226.

Grinblatt, M., and Keloharju, M., 2001. Distance, Language, and Culture Bias: The Role of Investor Sophistication. Journal of Finance 56, 1053-1073.

Hall, R.E., Jones, C.I., 1999. Why do some countries produce so much more output per worker than others? Quarterly Journal of Economics 114(1), 83–116.

Head, K., and Mayer, T., 2014. Gravity equations: workhorse, toolkit, and cookbook. In Gopinath, Helpman, and Rogoff (Eds.) Handbook of International Economics, Vol. 4, Elsevier.

Head, K., Mayer, T., and Ries, J., 2009. How Remote Is the Offshoring Threat? European Economic Review, 53(4), 429–44.

Houston, J., Lin, C., and Ma, Y., 2012. Regulatory arbitrage and international bank flows. Journal of Finance 67(5), 1845-1895.

Huberman, G., 2001. Familiarity Breeds Investment. Review of Financial Studies 14, 659-680.

IMF, 2022. Coordinated Portfolio Investment Survey (CPIS), https://data.imf.org/?sk=B981B4E3-4E58-467E sId=1424963554286.

Jeon, J. Q., and Ligon, J. A., 2011. The role of co-managers in reducing flotation costs: Evidence from seasoned equity offerings. Journal of Banking and Finance 35, 1041–1056.

Karolyi, G.A., 2006. The World of Cross-Listings and Cross-Listings of the World: Challenging Conventional Wisdom. Review of Finance 10, 1-54.

Kimura, F., and Lee, H., 2006. The Gravity Equation in International Trade in Services. Review of World Economics/ Weltwirtschaftliches Archiv 142(1), 92-121.

Krugman, P., 1980. Scale economies, product differentiation, and the pattern of trade. The American Economic Review, 70(5), pp.950-959.

Lane, P.R., and Milesi-Ferretti, G.M., 2005. The international equity holdings of euro area investors. Working paper, Trinity College and IMF.

Lane, P.R., and Milesi-Ferretti, G.M., 2008. International investment patterns. Review of Economics and Statistics 90(3), 538-549.

Ljungqvist, A., Marston, F., and Wilhelm, W. J., 2006. Competing for securities underwriting mandates: Banking relationships and analyst recommendations. The Journal of Finance, 61(1), 301-340.

Ljungqvist, A., Marston, F., and Wilhelm, W. J., 2009. Scaling the hierarchy: How and why investment banks compete for syndicate co-management appointments. Review of Financial Studies, 22(10), 3977–4007.

MSITS, 2010. The Manual on Statistics of International Trade in Services, https://unstats.un. org/unsd/tradeserv/TFSITS/msits2010.htm.

Manova, K., Foley, F., 2015. International trade, multinational activity, and corporate finance. Annual Review of Economics 7, 119–146.

Mayer, T., and Zignago, S., 2011. Notes on CEPII's distances measures: the GeoDist Database,

CEPII Working Paper 2011-25.

Melitz, M.J., 2003. The impact of trade on intra-industry reallocations and aggregate industry productivity. Econometrica, 71(6), pp.1695-1725.

Moody's, 2022. Sovereign Supranational, https://www.moodys.com/researchandratings/market-segment/ sovereign-supranational/-/005005?tb=0.

Nordås, H. K., and Rouzet, D., 2017. The Impact of Services Trade Restrictiveness on Trade Flows. The World Economy 40(6), 1155-1183.

Okawa, Y., and van Wincoop, E., 2012. Gravity in International Finance. Journal of International Economics 87, 205–215.

Papaioannou, E., 2009. What drives international financial flows? Journal of Development Economics 88(2), 269-281.

Papaioannou, G.J., and Karagozoglu, A.K., 2017. The Underwriting Business: Functions, Organization, and Structure. In Papaioannou, G.J. and Karagozoglu, A.K. (editors) Underwriting Services and the New Issues Market. Academic Press, 1-17.

Pichler, P., and Wilhelm, W., 2001. A theory of the syndicate: Form follows function. Journal of Finance 56, 2237–2264.

Pollock, T.G., Porac, J.F., and Wade, J.B., 2004. Constructing deal networks: Brokers as network 'architects' in the U.S. IPO market and other examples. Academy of Management Review 29, 50–72.

Portes, R., and Rey, H., 2005. The determinants of cross-border equity flows. Journal of International Economics 65, 269-96.

Portes, R., Rey, H., and Oh, Y., 2001. Information and capital flows: the determinants of transactions in financial assets. European Economic Review 45, 783–796.

PRS Group, 2022. The ICRG Methodology. https://www.prsgroup.com/wp-content/uploads/ 2022/04/ICRG-Method.pdf

Rischen, T. and Theissen, E., 2021. Underpricing in the euro area bond market: New evidence from post-crisis regulation and quantitative easing. Journal of Financial Intermediation 46, 100871.

Rodrik, D., Subramanian, A., Trebbi, F., 2004. Institutions rule: the primacy of institutions over geography and integration in economic development. Journal of Economic Growth 9(2), 131–165.

Sampson, G.P., and Snape, R.H., 1985. Identifying the Issues in Trade in Services. World Economy, 8(2), 171-82.

Santos Silva, J., and Tenreyro, S. 2006. The log of gravity. Review of Economics and Statistics 88, 641-658.

Sarkissian, S. and Schill, M.J., 2004. The Overseas Listing Decision: New Evidence of Proximity

Preference. Review of Financial Studies 17(3), 769-809.

Shipilov, A.V., 2006. Network strategies and performance of Canadian investment banks. Academy of Management Journal 49, 590–604.

Tesar, L., and Werner, I., 1995. Home Bias and High Turnover. Journal of International Money and Finance 14, 467-92.

UNCTADSTAT, 2021. United Nations Conference on Trade and Development database, https://unctadstat.unctad.org/wds.

Van Nieuwerburgh, S., and Veldkamp, L., 2009. Information Immobility and the Home Bias Puzzle. The Journal of Finance 64(3), 1187-1215.

Veronique, S., Benassy-Quere, A., 2006. Institutions and bilateral asset holdings. CEPII Working Paper 2006-19.

Vithanage, K., Neupane, S., and Chung, R., 2016. Multiple lead underwriting syndicate and IPO pricing. International Review of Financial Analysis 48, 193–208.

World Bank, 2022. Resolving Insolvency methodology, https://archive.doingbusiness.org/en/methodology/resolving-insolvency.

Wójcik, D., 2011. The global stock market: Issuers, investors, and intermediaries in an uneven world. Oxford University Press.

Wójcik, D., Knight, E., O'Neill, P., and Pažitka, V., 2018. Economic geography of investment banking since 2008: The geography of shrinkage and shift. Economic Geography 94(4), 376-399.

Walsh, K., 2008. Trade in Services: Does Gravity Hold? Journal of World Trade, 42(2).

Webster, A., and Hardwick, P., 2005. International trade in financial services, The Service Industries Journal, 25(6), 721-746.

Appendix

A Countries included in the sample

The countries included in the sample are Algeria, Argentina, Armenia, Australia, Austria, Bahamas, Bahrain, Bangladesh, Barbados, Belgium, Bermuda, Bosnia and Herzegovina, Botswana, Brazil, British Virgin Islands, Bulgaria, Cambodia, Canada, Cayman Islands, Central African Republic, Chile, China, Colombia, Costa Rica, Cote d'Ivoire, Croatia, Cyprus, Czech Republic, Denmark, Dominican Republic, Ecuador, Egypt, Estonia, Faeroe Islands, Fiji, Finland, France, Georgia, Germany, Ghana, Greece, Hong Kong, China, Hungary, Iceland, India, Indonesia, Iraq, Ireland, Israel, Italy, Jamaica, Japan, Jordan, Kazakhstan, Kenya, South Korea, Kuwait, Lao PDR, Latvia, Lebanon, Lithuania, Luxembourg, Macao SAR, China, Macedonia, Madagascar, Malawi, Malaysia, Mauritius, Mexico, Mongolia, Morocco, Myanmar, Namibia, Netherlands, New Zealand, Nigeria, Northern Mariana Islands, Norway, Oman, Pakistan, Papua New Guinea, Peru, Philippines, Poland, Portugal, Puerto Rico, Qatar, Reunion, Romania, Russian Federation, Rwanda, Saudi Arabia, Sierra Leone, Singapore, Slovak Republic, Slovenia, South Africa, Spain, Sri Lanka, St. Kitts and Nevis, Sweden, Switzerland, Syrian Arab Republic, Taiwan, Tanzania, Thailand, Togo, Trinidad and Tobago, Tunisia, Turkey, Uganda, Ukraine, United Arab Emirates, United Kingdom, United States, Uruguay, Vietnam, Zambia, and Zimbabwe.

B Robustness tests

We perform a couple of robustness checks for the results in column (2) of Table 1. First, we control for the degree of financial integration between the trading partners. While syndication ties are themselves a key feature of global financial market integration, the existing degree of integration might matter for trade in financial services beyond the formation of new ties. ²⁶ We augment Equation (2) with a proxy for financial integration, using two different proxies in turn. The first proxy is a measure of the equity asset holdings of the exporting country (underwriter) in the importing country (issuer) as a percentage of the total overseas equity asset holdings of the exporting country (Equity holdings_{jkt}). This proxy is constructed using data from the IMF Coordinated Portfolio Investment Survey (CPIS). The second proxy is a measure of the stock of lending ties between the trading partners as in Caballero et al. (2018) (Lending ties_{jkt}). These proxies are described in section 4.5. The results are in columns (1) and (2) of Table A1. In both cases, the coefficient on newly formed core syndication ties is robust to the inclusion of financial integration. The latter attracts a non-significant coefficient independent of the proxy we use.

 $^{^{26}}$ In the context of lending markets, financial integration has been shown to increase lending to commercial borrowers (see e.g. Berger et al., 2021).

	(1)	(2)	(3)
$s_{jkt}^{\mathbf{core}}$	$\begin{array}{c} 0.282^{***} \\ (0.0262) \end{array}$	$\begin{array}{c} 0.233^{***} \\ (0.0412) \end{array}$	$\begin{array}{c} 0.245^{***} \\ (0.0322) \end{array}$
Equity holdings $_{jkt}$	-0.000326 (0.00289)		
${\bf Lending} \ {\bf ties}_{jkt}$		-0.000138 (0.000770)	
$\mathbf{Ownership} \ \mathbf{ties}_{jkt}$			-0.00764 (0.00776)
Constant	17.79^{***} (0.0636)	$19.21^{***} \\ (0.298)$	$ \begin{array}{c} 19.14^{***} \\ (0.113) \end{array} $
Observations	9.758	11.156	14.997
Exporter-year FE (jt)	Yes	Yes	Yes
Importer-year FE (kt)	Yes	Yes	Yes
Pair FE (jk)	Yes	Yes	Yes
Number of exporters	47	67	67
Number of importers	60	96	97
Prop. of zero trade flows	0.492	0.509	0.539
Pseudo R^2	0.962	0.986	0.986

 Table A1: Robustness checks for Table 1

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01. Standard errors clustered on origin and destination countries.

Next, we turn to the issue of ownership ties. In some cases, an underwriter is a parent of a bank located in the issuer's country. If the issuer has a pre-existing lending relationship with the subsidiary, it may seem natural for a parent to underwrite the issuer's equity offering. This could mean that patterns of international trade in equity underwriting services might be driven by pre-existing lending relationships. Despite its apparent plausibility, this mechanism is very unlikely to be the only channel, or even the most important channel, explaining the underlying choice processes of issuers and underwriters, which ultimately translate into patterns of international trade. We can tell from our data set if the underwriter parent owns a subsidiary in the issuer's country. Underwriter parents own a subsidiary that is located in the same country as the issuer client for 45.7% of cross-border equity underwriting trade flows (42.4% when value weighted). Our data set, however, does not include information on pre-existing lending ties. Therefore, we cannot tell what fraction of underwriters had a pre-existing lending relationship with the issuers for whom they have underwritten an equity offering. We therefore reviewed the literature on the link between pre-existing lending relationships and the

choice of equity securities underwriters in order to assess the relevance of this channel for our analysis.

The literature finds that pre-existing lending relationships with issuers can give underwriters a small advantage in winning equity underwriting mandates. Bharath et al. (2007) find that "A bank that had been a lender in the past had a three-times higher probability of winning the future IPO underwriting business (increasing from 0.48% to 1.81%)" (p.411). This is, however, not the case for secondary equity offerings (SEOs). In this case, past lending relationships do not provide any additional benefit over and above what a prior underwriting relationship provides (Bharath et al., 2007). In addition, Ljungqvist et al. (2006) find that on average only 2.4% of loans were provided to issuers by their equity underwriters in the five years prior to their equity offering. The literature finds that prior underwriting relationships matter more than prior lending relationships. Indeed, a prior underwriting relationship between an underwriter and an issuer is one of the key factors influencing the odds of selecting a particular underwriter (Ljungqvist et al., 2006; Bharath et al., 2007). Crucially, underwriters that do not win a lead management mandate to underwrite an equity offering for an issuer can still participate in the issue by winning a co-management appointment in the underwriting syndicate. These appointments allow them to develop new underwriting relationships with issuers and win more lucrative lead management appointments in the future (Ljungqvist et al., 2009). Consequently, participation in underwriting syndicates creates a virtuous cycle whereby underwriters develop relationships not only with other underwriters, but also issuers, which ultimately gives underwriters better access to deals. In conclusion, despite some evidence that pre-existing lending ties can help underwriters win new business, in particular in IPOs, their economic significance is modest in comparison to prior underwriting ties with issuers (Ljungqvist et al., 2006; Bharath et al., 2007) and access to deal flow through underwriting syndicates (Lljungqvist et al., 2009).

The literature points to a very limited role for pre-existing lending relationships. We nevertheless perform a robustness check where we augment Equation (2) with a proxy for parent-subsidiary networks (Ownership ties_{jkt}). The construction of this variable is described in section 4.5. While this is not a direct measure of past lending relationships, it captures ownership ties between the parent underwriter and subsidiaries located in the country of the issuer. If past lending relationships are the main channel driving the results, the effect should show up in the coefficient on ownership ties. The results are in column (3) of Table A1. The coefficient on Ownership ties_{jkt} is insignificant.